

### Article The Politics of Market Change towards Sustainability: Revisiting Germany's Policy Support Framework for Renewables

Cristian Pons-Seres de Brauwer <sup>1,2</sup>

- <sup>1</sup> Society, Markets and Policy Group, DTU Wind and Energy Systems, Technical University of Denmark, Frederiksborgvej 399, 4000 Roskilde, Denmark; cpsb@dtu.dk
- <sup>2</sup> Environmental and Energy Systems Studies, Department of Technology and Society, Lund University, P.O. Box 118, SE-221 00 Lund, Sweden

Abstract: Legislative efforts for renewables-based energy decarbonisation hinge upon the support and commitment from different stakeholders holding often conflicting positions regarding disruptive processes of socio-technical transformation. However, the evolving acceptance of market actors on the policy-driven promotion of renewables over time remains under-scrutinised. Simultaneously, despite growing attention to power and politics in sustainability transitions, limited efforts remain invested for elucidating the political-economic nature of the market-based selection environments they are operationalised through, highlighting the need for a more systematic comprehension of the "politics of selection". To address these shortcomings, this paper provides a more refined understanding of the role of policy-driven markets and its participating agents in facilitating/hindering innovation diffusion and broader (system-wide) sustainability transitions. To do so, it showcases a longitudinal case study of the politics underlying Germany's evolving feed-in policy support framework for orchestrating a market-mediated diffusion of renewables (1980s-2020). Based on policy analysis and semi-structured interviews, the study traces the changing acceptance and ensuing strategic (re)actions of market actors to the emergence and evolution of Germany's market for electricity from renewable energy sources. Results show how different market participants effectively shape the selection environments they operate in by proactively contesting/deluding the design features of the support policies organising their economised relations (e.g., market entry conditions, exchange rules, remuneration levels, pricing schemes, etc.). Such efforts are undertaken through legal means and market framing strategies targeting the affordability of policy support costs, coupled with the strategic use of policy instrumentation as a vehicle to further expand/retain their market shares to the detriment of competing actors.

**Keywords:** market politics; policy acceptance; market-shaping; renewable energy; energy transition; actor diversity; selection environment

### 1. Introduction

Policymaking efforts for decarbonising national energy systems will require the sustained support, commitment, and ensuing participation (i.e., acceptance) from a wide range of actors holding disparate—and, more often than not, conflicting—positions with respect to disruptive processes of socio-technical system transformation [1–3]. This is particularly the case given the projected increase in renewable power capacity required to realise increasingly ambitious climate and energy targets [4,5]. A key driver of renewables-based energy decarbonisation has revolved around the deployment of policy support frameworks for incentivising a greater uptake of renewable energy investments and, in doing so, institute a market for electricity generated from renewable energy sources (RES-E market) [6].

Within a European context, Germany is exemplary of how "substantial support policies have effectively been pushing RES into the electricity market" [7] (p. 150). Its experience is illustrative of how the development and growth of a country's RES-E market is tied to



Citation: Pons-Seres de Brauwer, C. The Politics of Market Change towards Sustainability: Revisiting Germany's Policy Support Framework for Renewables. *Energies* 2022, 15, 3898. https://doi.org/ 10.3390/en15113898

Academic Editor: David Mares

Received: 6 May 2022 Accepted: 23 May 2022 Published: 25 May 2022

**Publisher's Note:** MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



**Copyright:** © 2022 by the author. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). that of the policies and instruments that it relies upon for its formation and steering [8]. However, the success of the country's feed-in policies for promoting renewables has triggered increasing opposition from electricity market incumbents (i.e., conventional energy utilities) against their continued use while advocating for policy alternatives better aligned with their corporate interests and market positions [9–12]. Their changing attitudes and ensuing responses towards different RES support policies highlights how policy acceptance is an irreducibly political endeavour, with shifts in the relative acceptance of different market actor groups empirically instantiating the evolving contestations, disputes, and conflicts underlying the legitimacy of the policy-driven yet market-mediated promotion of renewables.

Yet despite growing attention to power and politics in sustainability-oriented sociotechnical transitions [13,14], limited efforts remain invested to elucidate the inherent politicaleconomic nature of the market-based processes they are operationalised through [15]. In this respect, the sustainability transitions literature has yet to challenge its apolitical and valuefree treatment of markets, contributing instead a more systematic understanding of the "politics of selection" [16,17]. This requires a more thorough evaluation of what (and why) different policy support mechanisms work and others do not [18], which consecutively demands a more elaborate appreciation of the various strategies pursued by market participants to challenge, alter, and/or disrupt (i.e., shape) the policy-driven (re)configuration of the selection environments they operate in for orchestrating change in structures of energy production and supply [19,20]. A greater scrutiny of these contentious marketshaping practices would in turn contribute a response to recent calls for examining the role of markets and their participating agents in facilitating, expediting, and/or hindering policy-driven energy system transformation towards sustainability [21].

Against this backdrop, this paper revisits Germany's transition towards a renewablesbased electricity system to examine—by means of a longitudinal case study—the politics of policy-driven market reconfiguration processes purposively enacted to realise sustainability objectives. To do so, a review of recent market-shaping literature with political analyses of markets is first conducted to then outline an analytical framework-foregrounded on a sociological perspective of markets-for tracing changes in the "acceptance positions" (e.g., opposition, concern, reluctance, indifference, dismissal, consent, support, etc.) of market actors in relation to the country's evolving feed-in policy support framework for orchestrating a market-mediated diffusion of renewables. This calls for an explicit focus on RES policy support instrumentation "insofar as the increase in [renewable power] capacity and investments has been driven by active policies to this end" [22] (p. 121). Such seems to be the case in Germany, whose RES-E market emerged and developed due to the prolonged use of feed-in policy instruments for supporting the market diffusion of what once were functionally promising yet commercially vulnerable (i.e., market immature) socio-technical innovations such as wind and solar power [23–27]. Therefore, elucidating the politics underlying the emergence and evolution of Germany's RES-E market inevitably entails a necessary examination of the changes undergone by the very same feed-in policies that enabled its constitution, organised its configuration, and steered its development in the first place.

To this aim, this study addresses the question of how have different market actors contested and influenced the evolution of Germany's feed-in policy support framework for RES-E generation: to what extent have their vested interests been reflected in the (re)formulation of feed-in policy support instrumentation, and how have such market-shaping efforts affected different market participants' capabilities to partake in the RES-E market? The analysis uncovers how different market actors effectively shape the selection environments they operate in by legally contesting, framing, and misusing the very same policies organising their economised relations and, by extension, steering the actor's configuration of the RES-E market. It shows how such efforts are undertaken through legal disputes against specific policy design features, coupled with the use of discursive strategies elevating their cost-effectiveness and challenging their affordability while downplaying

their socio-ecological benefits. Ultimately, the prevailing use of market-shaping practices, if left unchecked or unaccounted for in the design of RES policy support frameworks, might eventually skew RES-E market participation in favour of a reduced number of large market players with greater financial and operational endowments to operate in selection environments with the overriding concern of economic efficiency. Policymaking efforts for renewables-based energy decarbonisation should therefore assess the net socio-ecological welfare outcome resulting from the relative benefits of economic efficiency along with the environmental effectiveness and distributive impacts of RES policy support instrumentation, reflecting a more balanced application of efficiency, pace, and quality considerations in a policy-driven yet market-mediated energy transition.

Following this introduction, Section 2 reviews the market-shaping and -politics literature informing a policy-centric analytical framework for examining the politics of sustainability-oriented market change. Section 3 describes the methodological approach adopted. Section 4 analyses a longitudinal case study of Germany's RES-E market, consisting of a stage-wise chronological account of the changes undergone by its feed-in policy support framework from the 1980s to 2020. Building on the identified changes in policy support instrumentation, Section 5 discusses a number of insights cutting across market politics, sustainability transitions, and policy acceptance. Section 6 concludes, outlining limitations and suggestions for future research.

#### 2. Politics of Market Change towards Sustainability

#### 2.1. Market-Shaping: A Process of Changing Economised Relations across Value Networks

A sociological perspective of markets departs from simplistic notions of "the market" as one single, naturally given structure of efficient resource allocation dis-embedded from society [28,29]. Instead, it elaborates a more empirically sophisticated appreciation of "markets" as malleable socio-technical configurations organising economised relations across networks of value creation and exchange [30–32]. Such relations are established between a heterogeneity of market agents with different interests and resources across the value network. They include market participants like producers, suppliers, traders, retailers, and consumers, yet also comprise other actor groups holding a stake in the market (i.e. concerned with the correct functioning of its institutional and functional architecture, its performance, its outcomes, etc.) such as policymakers and regulators, industrial lobbyists, consumer associations, financiers, etc.

Appraising markets as socio-technical configurations embraces the co-existence of market actors alongside different material "actants" such as technologies, discourses, policies, etc. [33]. In light of actors' respective attempts to orchestrate market interaction, these material actants or "market devices" are articulated and deployed in order to perform certain market-shaping functions towards specific objectives [34]. As such, rather than being (mis)understood simply as reactive entities adapting to changes in a selection environment that they cannot influence or modify, firms and other market actors are instead conceived as proactive agents shaping multiple selection features of market environments in order to (re)organise patterns of economised relations that advance their interests [17,35–37].

"Market-shaping" is thus understood as purposively-enacted strategies through which economised relations for value creation and exchange are organised, contested and subverted, and reorchestrated in light of differing—and potentially conflicting—interests. In that respect, a more granular attention to the reiterative re-composition of market-shaping devices provides the opportunity to empirically explore purposive change of socio-technical market configurations towards sustainability [38,39]. Understanding such a process there-fore stands as a prerequisite to assess the relative performance of market-shaping efforts to induce socio-technical change towards more sustainable value networks of economised relations—or markets in transition.

#### 2.2. "Market-Shaping" Policy Devices

A sociological perspective of markets in transition directs the analytical enquiry to the articulation of strategic interests into market-shaping devices deployed in order to induce or steer economised relations (i.e., market interactions) that fulfil specific sustainability objectives [39,40]. These do not usually emerge naturally from incumbent market practices, but instead tend to be articulated by policy instruments from other market shapers (i.e., policy-makers, regulatory agencies) in order to induce societally desirable market interactions that would otherwise not occur without their introduction [17,32,41,42]. Sustainability-oriented markets therefore tend to be policy-driven markets: markets purposefully designed and steered by state authorities to realise sustainability goals such as renewables-based energy decarbonisation [39,40,43–45].

Policy-driven markets for sustainability have been articulated through multiple different market-shaping policy devices such as "production quotas" for more sustainable fisheries management in Norway [46], "certification schemes" for organically-grown coffee in Uganda [39], "carbon credits" traded across global greenhouse gas (GHG) offset markets [47,48], or "environmental standards" for sustainable biofuel production in the European Union (EU) [49]. Approaching Germany's evolving feed-in policy support framework for renewables as a chain of changing market-shaping policy devices therefore provides a resourceful means to scrutinise the process by which sustainability-oriented economic policy/legislation (e.g., StrEG, EEG) is articulated by concrete support instrumentation (e.g., feed-in-tariffs, market premia, auctions) aimed at shaping market (inter)actions (e.g., increased RES investments) towards the realisation of societally desirable climate and environmental goals (e.g., GHG emission reductions).

Scrutinising this process of articulation requires an empirically rich description of markets to elucidate *how* the combination of institutional arrangements and discursive frames are assembled into market-shaping policy devices to orchestrate sustainability-oriented market change, as well as *who* partakes in such assemblage [50]. Highlighting the "how" and "who" brings issues of power and politics in the (re)configuration process of market-based selection environments to the forefront of the empirical enquiry [35,51]: who is capable or allowed to participate in or influence the assemblage of the market-shaping policy device and who is not; and therefore, whose interests are better reflected in the (re)configuration of the market and whose are not?

#### 2.3. Market Politics

Policy-driven markets for sustainability are not exempt from political contestations [52]. Typically, these are brought forward by those market participants whose dominant position is threatened by the reconfiguration of current market arrangements [42,45]. Incumbents therefore have a strong incentive to politically mobilise to maintain the legal-regulatory framework upholding the established market configuration and impede its replacement with less favourable alternatives [51,53]. Market incumbency thus contributes to locked-in and path-dependant socio-technical configurations displaying a certain market inertia that is difficult to overcome [54–57].

Yet the same can be expected from other market participants (e.g., newcomers) who operate in unfavourable selection environments, as they may have an equally strong motivation to collectively influence policymaking efforts to replace the current legal-regulatory framework with the objective to unlock the market's incumbent-dominated configurational set up [42,51,58,59]. A political analysis of markets thus moves past the legally sanctioned practices of market participants (e.g., cost-reductions through scale economies, creative marketing campaigns, product diversification strategies, etc.) to uncover the contentiousness of market-based selection environments instantiated across different market-shaping actions.

Examples are plentiful and include, for instance, wholesale market traders benefitting a few large grain producers by setting minimum "production quotas" too high for small farmers to meet and access government procurement schemes in India (*market gatekeeping*) [60]; Japanese firms overtaking incumbent consumer electronics manufacturers in the

UK through strategic greenfield investments exploiting government "investment subsidies" for industrial development (*market usurpation*) [61]; auction participants colluding to secure themselves winning bids (and exclude competing bidders failing to cooperate) from telecommunication "auctions" in Europe, (*market co-optation*) [62]; European and US coalitions of solar PV panel manufacturers lobbying policymakers to impose "antidumping tariffs" on panel imports from Chinese competitors (i.e., *market protection*) [63,64]; or incumbent electric utilities shaping public opinion against a nuclear power "phase out scheme" in order to align Swiss voters' policy preferences with their corporate interests (*market framing*) [65].

Such instances shed light on the pervasiveness of contentious market-shaping practices exercised by market participants in their respective attempts to secure control over market opportunities and advantages claimed by competitors [51,66], a process that [67] defined as *social closure*. As such, they reflect the inherently political nature of selection environments as spaces of controversy, dispute, and conflict populated by market actors purposively steering the market's legal-regulatory framework in their attempts to extract and retain *influence rents*: the capture of unearned marginal value by re-designing and/or perverting the "rules of the game" to one's favour and the detriment of others [51,68].

To this end, these practices illustrate the strategic (mis)use of market-shaping policy devices as a means to enable/constrain the capacities of other actors to partake in economised relations and, by extension, re-configure the constellation of market participants. In doing so, market-shaping policy devices originally assembled for the market-mediated realisation of politically determined sustainability objectives (e.g., decarbonised energy supply) are effectively repurposed to operate as *tools of inclusion/exclusion* determining the ability of different actors to participate in the market. Success in undertaking such efforts leads to the reproduction of asymmetric power relations between market participants with unequal material, financial, and operational endowments to shape the selection environments they operate in [51,69].

Given the inherently political nature of market selection environments, and in the context of policy-driven market change for sustainability, a political appraisal of the emergence and evolution of Germany's RES-E market directs the analytical exercise towards (a) the contentious market-shaping practices of gatekeeping, usurpation, co-optation, protection, framing, etc., undertaken by different market actors operationalised through (b) the strategic contestation, (mis)use and re-configuration of the feed-in policy support framework's various design features (e.g., eligibility criteria, allocation mechanism, valuation method, remuneration levels, pricing scheme, duration, degression rates, etc.) with the objective to (c) include/exclude competitors from RES-E market participation (closure) and retain influence rents (e.g., market shares).

#### 3. Methodology

#### 3.1. Research Design

The emergence and evolution of Germany's RES-E market is appraised from the situational perspective of the feed-in policy support framework itself. Doing so enables to reposition the analytical standpoint from an "outsider" onlooker to an "insider" observer [70,71], affording a singular positional view from where to trace market actors' changing "acceptance positions" as reflected in their market-shaping practices with respect to the policydriven (re)configuration of market selection environments. From this singular standpoint, an in-depth case study is carried out of Germany's changing feed-in policy support framework for orchestrating a market-mediated diffusion of renewable energy technologies (RETs). The analysis encompasses the period between the late 1980s until 2020.

The research reported herein therefore consists of a longitudinal case study, a recurrently employed research design in the sustainability transitions literature, yet less extended across its market-shaping and acceptance counterparts [72–74]. This has prompted recent calls for conducting multilevel, longitudinal accounts within both groups of literature [75–77]. A longitudinal research design seems a suitable candidate for this case in point, as it allows

to qualitatively unravel and sequence complex phenomena unfolding over time within a given context [78–80]. As such, it offers a particularly supportive means to elaborate the process character of sustainability-oriented market evolution, as it favours empirically rich illustrations of policy-driven market-change dynamics unfolding across different stages along a chronological process [73,81]. Adopting a processual perspective on market change therefore facilitates the identification of long-term market-shaping processes while helping to recognise the evolutionary character of markets as both complex and dynamic socio-political constructs [73,82,83].

#### 3.2. Case Selection

Germany has been repeatedly depicted as representative of Europe's governance approach for addressing its grand societal challenges such as the Eurozone crisis, EU-Ukraine-Russia relations, or the climate crisis [84]. On the latter, Germany has often been catalogued as a "frontrunner" and a "leader" in its efforts to mitigate climate change [85–87]. Germany's "transition" efforts in decarbonising its energy system based on the mass-scale diffusion of renewables therefore lends itself as a suitable case with generalisable potential informing other parliamentary democracies with a market-based economy and robust industrial base in need of drastic decarbonisation measures. This is particularly the case for those 26 countries with which it shares a common climate and energy governance structure overseen by the European Commission (EC).

The choice for examining the German experience thus follows an "informationoriented" selection logic of a "typical" case study, where the aim is to maximise the utility of information from a single case of strategic relevance in its representability of the phenomena under scrutiny [88,89]. Given the process-character of the longitudinal research design, the selection logic is further driven by the choice of a "pathway" case study useful for elucidating causal chains when tracing empirical processes over time [88].

#### 3.3. Data Collection and Analysis

Multiple data sources were drawn upon to build the richness of data required for elaborating a detailed descriptive account of the case under examination [90]. Data collection started with the retrieval of secondary data sources. These consisted of, first, a number of historical studies on the various legislative developments unfolding across differing time periods. These studies consisted of academic literature as well as policy reports. They were identified through a keyword search in journal databases, followed by a snowball sampling procedure whereby relevant references in the consulted texts were taken for further review [91]. Additional secondary material was collected in the form of legal documentation (e.g., EU directives, national laws, parliamentary proceedings), white papers and roadmaps, market reports, and industry position papers, as well as quantitative data on various variables (e.g., investment flows, ownership structures, subsidy volumes, electricity prices, installed capacities, etc.). Data retrieval was conducted via online legal archives, EU/national statistical databases, and media and corporate repositories. With this, the various legislative developments, events, diffusion trends and milestones, as well as the market-shaping actors involved, were chronologically mapped and arranged into a linear progression of delineated timelines unfolding in parallel.

Secondary sources were complemented with primary data obtained from 19 semistructured interviews with different energy industry associations, policymakers and regulatory agencies, electricity market participants, and expert market analysists/consultants (Table 1). The first round of interviews was conducted between March-December 2020 followed by a second round in March 2022. Interviews lasted about one hour. They were all conducted online, audio-recorded, and transcribed. The transcription exercise enabled an initial analysis of the interview data, which was then further organised according to the delineated timelines previously constructed, and further catalogued according to the contentious market-shaping practices identified in the literature review. The ensuing analysis enabled to deepen some of the most salient policy events, conditions, and measures identified during the initial longitudinal appraisal. The collection and analysis of the interview data therefore served to corroborate, contest, nuance, expand, and/or discard the empirical material previously obtained, which deepened the exploratory character of the analytical exercise.

**Table 1.** Categorisation list of informants interviewed for the study, along with the number of interviews conducted for each category.

Informant *	Category	No Interviews
Renewable energy association	<b>RES</b> Association	2
Energy industry association	E-Association	1
Citizen energy association	C-Association	2
Policymaker, regulatory body, public agency	Government	4
Energy utility company	Utility	2
RES operator/developer	Independent power producer (IPP)	3
Energy cooperative	Cooperative	2
Energy market analyst/consultant	Consultant	3

\* Interviews are cited using the category of informant and a number (e.g., Government 1).

From this, a detailed historical plotline was elaborated in chronological order. Analytical attention was invested in richly describing key RES policy support scheme developments and exploring their coupling with the various market-shaping actions undertaken by market participants with a vested interest in their assemblage and steering, enabling a context-bound examination of evolving market  $\leftrightarrow$  policy  $\leftrightarrow$  society interactions.

#### 4. Longitudinal Case Study

# 4.1. Market Gatekeeping through "Voluntary" Compensation Schemes as Tools of Market Exclusion—Pre-1990

Before 1990, Germany lacked a nationwide (i.e., federal) regulation of RES-E compensation. Market access for renewables was not high up in the political agenda. Rather, privately negotiated arrangements were considered a sufficient means to organise the relationship between RES-E generation and grid access within the prevailing monopolistic organisation of the electricity market. RES-E could only be fed into the grid via "Voluntary Association Agreements" (VAAs), non-binding compensation frameworks subscribed on a voluntary basis between RES producers and energy utilities—in their role as grid operators [92]. Since the utilities owned both the electricity grid and the large nuclear/coal power units under vertically-integrated regional monopolies, they held a dominant position in the negotiation of the compensation regime for RES-E, enabling them to control entry to the electricity market (RES Association 1). Market gatekeeping was therefore the predominant closure strategy to retain influence rents from the prevailing monopolistic organisation of the electricity market. This was reflected in the method used to remunerate RES-E, which was based solely on the utilities' avoided generation costs [93].

The "avoided cost" principle for RES-E valuation resulted in very low payment rates insufficient for full cost recovery. For instance, the VAA between the Association of Bavarian Hydropower Plants and the regional grid operator established a remuneration of about EUR 0.4 ct/kWh for electricity generated by small and medium-sized hydropower plants, "less than what the electricity companies had to spend on their own electricity" [94] (para. 4). Similarly, the remuneration offered by a large utility company for a citizen's wind turbine consisted of no more than EUR 0.1 ct/kWh, while simultaneously charging EUR 0.14 ct/kWh for grid access [95]. Since VAAs were not strictly binding for either stakeholder, but just a recommendation, energy utilities could arbitrarily obstruct or entirely refuse RES-E fed into the grid by independent competitors (Consultant 1) [96]. Market incumbents therefore employed VAAs as tools for market exclusion to prevent newcomers from partaking in the electricity market. "The utilities were free to accept or not to accept, and this became a burden to upcoming renewable energy producers" (RES Association 1).

Inevitably, RES operators—mostly consisting of farmers and individual citizens—were effectively displaced towards the margins of an electricity market locked into the incumbency of a fossil/nuclear energy monopoly.

VAAs thus codified a patchwork of discriminatory relationships between electricity market incumbents (energy utilities) and newcomers (RES operators), with the latter facing significant risks due to unreliable investment conditions (i.e., undercompensation, non-guaranteed and overcharged grid access) and limited regulatory oversight. Not surprisingly, RES-E was rarely fed into the grid but mostly used for private consumption (Consultant 1). Its contribution to the electricity generation mix remained marginal throughout the 1980s. By 1990, it accounted for just 3.4% of gross electricity generation (Figure 1), with the majority (92%) coming from large hydropower plants owned by the utility companies [96,97]. Given the limited prospects of change stemming from such a stagnant market configuration, the introduction of a support scheme to promote renewables "seemed next to impossible" [98] (p. 10).



**Figure 1.** Shares of renewable power sources in gross electricity generation, 1990. Source: Own elaboration. Data from [97].

However, in 1989 the Ministry of Research launched a 15-year Research and Development programme to deploy 250 WM of wind power capacity [99]. To do so, it simulated a price-based remuneration scheme consisting of a supplement of EUR 4.10 ct/kWh added onto the negotiated tariff rate of EUR 4.4 ct/kWh, yielding a final remuneration of EUR 8.5 ct/kWh for wind-based electricity granted for 10 years [100]. Besides its core objective of monitoring the technical viability of wind power technology, the programme performed a perhaps less salient, yet crucial role as a "testing device" for assessing the performance of FiTs in orchestrating market interaction around RES-E. Their use "was legitimised by the need to gain practical experience with different [support] approaches under real-life conditions" [24] (p. 106), laying the groundwork for later-stage renewables' support schemes.

#### 4.2. Inception of Renewables' Support in a Nascent RES-E Market-1990 to 1999

The Feed-in Act of 1991 (*Stromeinspeisungsgesetz*—StrEG) constituted Germany's first legally-binding regulatory framework to support renewables. The law framed RES support as a response to increasingly salient climate and energy concerns held across the German public [101]. "The main trigger for political action was definitely a societal push. There was a very strong anti-nuclear movement in Germany at that time due to the Chernobyl nuclear disaster [of 1986]" (RES Association 1). RES policy support was therefore built upon an increasingly concerned population for environmental and climate issues [24,27,102], rather than exclusively stemming from any demands by market newcomers (RES operators).

The EC initially accepted the law due to its small financial endowment and limited impact on electricity prices [103]. Its position was paired with the utilities' dismissal of a support scheme "likely to average less than 0.1% of [their annual] revenues" [104] (p. 4)

(Consultant 2), enabling them to retain the influence rents of a monopolistic energy system. The market incumbents thus echoed their parliamentary counterparts in downplaying the law as "a treat for the ecologically motivated" [94] (para. 8), claiming that renewables would never supply more than 4–6% of Germany's electricity demand and wind energy would be less than 1% (Government 1, RES Association 1, Consultant 2) [105,106].

Building on the "250 MW Wind" programme, the StrEG's support framework combined compulsory grid connection with preferential dispatch for RES-E. In doing so, it undermined the incumbents' closure strategy of market gatekeeping, as it effectively forced energy utilities to allow market entry for new participants (Consultant 1). Yet their efforts to limit market entry for newcomers remained relatively uninterrupted throughout the 1990s and 2000s, as they continued to deny grid access to independent RES operators on the basis of grid capacity constraints and ensuing blackout risks (Cooperative 1) [107].

The support framework further obliged utilities to remunerate RES operators for the electricity fed into their grid area at a fixed percentage of the annual average electricity price (e.g., 90% for wind/solar—Figure 2). Tariff rates were therefore pegged to the electricity price, and thus, indirectly linked to the utilities' avoided generation costs (Consultant 1) [108]. Importantly however, avoided external costs of EUR 3–5 ct/kWh from coal power were accounted for when setting the remuneration level [109]. Due to this monetary endowment, the compensation scheme restricted RES operators' exposure to the arbitrariness of energy utilities in the negotiation of a cost-recovering remuneration scheme for RES-E, and thus counterbalanced (yet did not overthrow) their dominant position in their economised relations with renewables' producers (IPP 1-2). By setting avoided external costs as the benchmark for calibrating remuneration levels, FiTs levelled the playing field between RES-E and conventional electricity producers, itself a declared objective of the StrEG [101].

The provisions on guaranteed grid access and purchase obligations effectively created a steady and (to a lesser extent) predictable demand for RES-E (IPP 1-2). Importantly, in accounting for the avoided external costs of conventional energy generation, the support framework established a differentiated market space that, thanks to an environmentallyreflective remuneration and preferential treatment for RES-E, organised market exchanges upon the basis of criteria differing from the prevailing economic ordering of the electricity market. Its emergence was furthered by explicitly restricting incumbents from partaking in it, as the eligibility criteria was purposefully articulated to exclude energy utilities from remuneration [101].

The combined performance of these design features triggered a notable increase in wind power capacity undertaken by a growing number of new market participants. These consisted mostly of community-based models of wind turbine ownership (Government 1, RES Association 1): in 1989, citizen-owned wind turbines numbered 221 across Germany, and by 2001, they had increased to 9359 [110,111]. Neither solar PV nor biomass experienced significant growth due to prohibitively high technology costs and remuneration levels too low to ensure cost-recovery despite "auxiliary" support such as the "1000 Rooftops" programme for solar PV) (Government 1). Virtually none of the wind power capacity installed during that decade was owned by the incumbent utilities [96].



**Figure 2.** StrEG remuneration levels for wind and solar power (set at 90% of the average electricity price) along with the annual growth rate for onshore wind power capacity. Source: Own elaboration. Data from [24,97,112].

Yet the policy-driven diffusion of renewables did not unfold without the contestation of some market incumbents who saw their position potentially at risk by an increasingly actordiverse market configuration (Government 1, RES Association 2). "From the utilities' point of view, this notion of independent power producers was a threat because that meant more players in the marketplace, and the utilities just didn't want their monopoly threatened" (RES Association 1). Their closure strategy of market gatekeeping progressively expanded to include legal contestation. Exemplary of such efforts was their judiciary actions taken at national and EU levels, based on the StrEG's discriminatory nature for lacking a burdensharing mechanism to distribute remuneration costs equally across all end-users. Utilities operating grid areas with a higher RES-E penetration than others would have to accrue higher costs than their competing suppliers, resulting in a competition-impairing outcome incompatible with the EU Internal Electricity Market [113]. The core dispute targeted "the calculation of the minimum purchase price for electricity generated from wind" [114] (p. 6), which allegedly "inflated" the avoidable costs of conventional electricity generation by EUR 4.3 ct/kWh—the same cost value used in the "250 MW Wind" programme and within the value range employed in the StrEG to account for the external environmental costs of coal power. Arguably, if the remuneration level for wind energy remained unchanged that is, if external environmental costs remained accounted for-there would be "a risk of overcompensation with the ensuing detrimental effects on competition" [114] (p. 6). The outcome of the incumbents' closure strategy by legal means yielded mixed results: the demands to lower the 90% purchase price for wind were side-lined by the government, who instead capped at 10% the amount RES-E utilities had to feed into the grid—effectively limiting the future growth of a nascent RES-E market.

Furthermore, the liberalisation of Germany's electricity market in 1998—while beneficial for new participants due to the opening up of a market monopoly to competition meant that electricity prices were now determined in a competition-driven marketplace, which resulted in price reductions thereafter (Figure 2) (C-Association 1). Since tariff rates were pegged to retail electricity prices, the remuneration obtained by RES-E producers dropped along with the certainty of their revenues (Consultant 1) [115]. These were further eroded by the impossibility of confidently foreseeing RES-E demand, as the StrEG did not specify a duration of support but instead granted it on a yearly basis.

# 4.3. Emergence and Consolidation of a Competition-Exempt RES-E Market: Market Usurpation through FiTs as Tools of Market Inclusion—2000 to 2010

The StrEG's faulty design to address liberalisation-induced market changes and the energy utilities' closure strategy by legal means combined to undermine the emergence

of a still unstable RES-E market (mostly driven by wind power) now at risk of losing momentum. "A lot of wind investment feared bankruptcy or insolvency. So, we had big demonstrations demanding higher compensation" (Government 1). The government thus "had to act fast to restore a secure climate for investment [...] as more and more renewable energy investment projects were being shelved" [116] (p. 3). This regulatory uncertainty found its resolution in the Renewable Energy Sources Act of 2000 (*Erneuerbare Energien Gesetz*—EEG).

The EEG reflected the political resolve of a pro-renewables Red/Green government coalition rather than the sole outcome of a cohesive associative action from market newcomers to secure effective (i.e., sufficient) and sustained (i.e., predictable) remuneration (RES Association 1). Like their 1990 predecessors, policymakers leveraged widespread public support for climate action to orchestrate a pro-RE alliance backing FiT-based renewables' support (RES Association 1). "The basic motivation came from a societal demand, not from renewable businesses. [They] only begged for better wind power compensation, nothing else. They did not beg for photovoltaics, hydropower, or biogas. We had to organise it. We had to force companies to support FiTs for all technologies" (Government 1). Without actively pursuing it, market newcomers found in their legislative counterparts a cohort to avail themselves participation in a utility-dominant electricity marketplace.

The law encoded an ambitious remuneration scheme based on the full cost-range of RETs. It substituted the "avoided external costs" valuation benchmark and instead calibrated remuneration levels based on the specific cost structures of each technology. From 2000 onwards, RES-E valuation was performed via fixed FiTs operating independently from demand-driven electricity price developments, effectively shielding RES operators from the increased price volatility of a liberalised electricity market. Their use represented a paradigm shift in the operationalisation of Germany's support framework: FiTs guaranteed payment for every kWh of RES-E fed into the grid regardless of its actual market need (i.e., demand) for a total duration of 20 years (Utility 1, Government 2). The combined performance of these design features did not necessarily align Germany's RES-E market with a competition-based model of economised relations, but rather, bypassed it to enhance revenue certainty for RES operators, and in doing so, ensure the investability of renewables as a means to operationalise ambitious climate and environment goals demanding at least a 12.5% market share by decade's end (Utility 1, Government 2, IPP 1) [116].

Additional provisions were articulated to redress the faulty design features of discriminatory market access and cost-defective market exchange previously exposed by the incumbents' legal contestations against the StrEG. First, the eligibility criteria penalising energy utilities was replaced with an "actor neutral" scheme guaranteeing support for any RES operator. This provision equipped FiTs to perform as tools of market inclusion, since utility companies were now eligible for FiT-based remuneration. Since they were no longer excluded from the RES-E market, the incumbents lost one key point of contention in their closure strategy to delegitimise the support scheme upholding its emergence. As argued by Government 1, "when we exclude the big old utilities from receiving support, we have them every time as an enemy. We must allow them to make their business case with this law".

Second, to enhance positive market spillover (e.g., cost reductions through technological innovation), and later, to contain detrimental ones (e.g., windfall profits from overcompensation), the remuneration scheme incorporated scheduled tariff degressions (Consultant 1). This was complemented by a novel "equalisation scheme" distributing support costs equally across all electricity end-users through a levy on their electricity consumption. However, exemptions for energy-intensive industrial consumers were granted to maintain their competitiveness, a provision that would yield important spillover and trigger cost-containment efforts later on (Government 1–3, RES Association 1). By introducing a burden-sharing mechanism for "equitable" support costs distribution, the EEG rendered obsolete the 10% cap imposed on RES-E fed into the grid. Energy utilities burdened with higher RES-E shares were no longer disadvantaged with respect to their competing suppliers, a second element which further weakened their legal manoeuvres against a "competition-impairing" allocation of support costs. Despite their opposition to FiT-based support due to its "exorbitant costs" [115], their closure strategy was ultimately derailed with a favourable court ruling on the legality of Germany's feed-in policy support framework [117]. Shortly afterwards, the EC dropped its objections against the EEG, assessing later that "FiT regimes are generally the most efficient and effective support schemes for promoting renewable electricity" [118] (p. 3). Market incumbents thus lost an important political lever for their closure efforts, as the EC's position with respect to RES policy support had traditionally favoured a tradeable renewable certificates scheme for orchestrating renewables' support [6].

This however did not undermine the incumbents' stronghold on the electricity market, as their remaining role in grid operation allowed them to maintain their market gatekeeping efforts to slow down the market diffusion of RETs. As explained by Cooperative 1, exemplary of such efforts were:

"Some utility companies [who] denied grid access for many high-capacity PV installations. Even though the [EEG] law says they must grant them access, they just didn't. They said 'yeah, file a lawsuit', and when they did the utilities won almost every case. Plus, many solar PV owners didn't have the time or money to pay for this, so instead they just built smaller installations. This aggressive behaviour tells you a lot about how they act, to bully out the small participants from the market".

At the same time, larger utilities consolidated their market power by leveraging liberalisation reforms to buy out competitors and absorb their supply areas. Their market expansion strategy of fast and profitable "growth by acquisition" led to an actor consolidation resulting in an oligopolistic market configuration dominated by four large power companies—RWE, E.ON, EnBW and Vattenfall [9]. By 2004, the "Big 4" owned 82% of Germany's electrical power capacity, controlled the entire transmission infrastructure, and almost all distribution networks [92,119]. In 2009, they still accounted for 80% of Germany's total electrical capacity and generated 82% of the country's electricity [120]. On the other hand, between 2004–2010 they accounted for just 18% of the growth in biomass capacity, 2.7% in onshore wind, and 0% in solar PV and biogas [121]. By decade's end, they operated just 6.5% of the country's renewable power portfolio, mostly from large hydropower plants [121].

"The utility companies did not invest [in renewables] because of one thing: FiTs enabled annual returns of about 5–7%, and they had a thinking of 20% returns. It was too small a profit" (Government 1) [122]. Market incumbents thus forewent their participation in the RES-E market, giving it away to an increasing number of new market actors who seized the opportunity provided by long-term, fixed FiTs to secure a footing in RES-E generation (Figure 3) (Government 2, Consultant 1). In doing so, they began a process of market usurpation of an incumbent-dominated electricity generation base.



**Figure 3.** Evolution of the ownership structure of renewable power capacity in Germany between 2004–2010. Source: Adapted with permission from Ref. [121]. 2011, trend: research.

Between 2000–2010, wind power quadrupled its production, biomass electricity generation increased by a factor of 13, biogas by a factor of 35, and solar PV by a breath-taking factor of 195 [97]. The 2010 target of 12.5% market share was amply surpassed by 2007, reaching a 17% share by decade's end [97]. The prolonged use of FiTs throughout the 2000s therefore drove the emergence of an actor-diverse RES-E market configuration, itself spearheaded by a drastic increase in citizen-led developments via cooperative associations (Figure 4) (RES Association 2, C-Association 2) [123]. Germany's RES-E market became characterised by diversity in the generation base—both in terms of power capacity and actor configuration—and by concentration in its supply and distribution segments (RES Association 2).



**Figure 4.** Evolution of the number of energy cooperatives in Germany between 2000–2017. Source: Adapted from [124].

Yet the emergence of a policy-dependent RES-E market did not occur without its challenges. By decade's end, climate action progressively ceased to play a dominant role in framing RES promotion while rising support costs became an increasingly salient concern dominating public and political debates over Germany's FiT-based support framework (Government 4) [125,126]. In effect, the expansion of renewables throughout the 2000s had been accompanied by a steady increase in the EEG levy for end-use electricity consumers—from EUR 0.2 ct/kWh in 2000 to EUR 3.5 ct/kWh in 2011 [127]. Simultaneously, RETs had undergone steady cost reductions during the same time period, rendering RES-E generation increasingly competitive in a competition-exempt RES-E market (Utility 1) [128].

Rapidly declining costs fuelled political concerns over windfall profits from overcompensation as well as from rising household electricity prices (Government 4) [129,130]. "At some point the government was over-subsidising RES installations with this feed-in regime" (Utility 1). "To some degree this is true because FiTs were quite high until 2012, so you could get easy returns. So yeah of course there was a profit-seeking motivation" (Cooperative 2). Market newcomers' early-day environmental idealism progressively gave way to rent-seeking pragmatism: instead of reacting to contain these detrimental market spillovers, they opted to retain their influence rents by demanding better remuneration conditions (IPP 1). Yet some of them adopted a more strategic stance and argued for tariff reductions for fear of legitimacy losses from rapidly increasing support costs [115]. As reflected by IPP 1:

"Around 2008–2009 there were a lot of claims to reduce FiTs. So many renewable energy associations, companies and so on came along asking politicians to not reduce them, but to increase them. Most of that to be honest was avarice. They just wanted even bigger

margins, so they asked for higher and higher FiTs. What should have been implemented initially is a serious reduction of FiTs. Here is the one key mistake the renewable energy players made".

### 4.4. Market Framing and Co-Optation: The Politicised Use of Support Costs and Auctions as Tools of Market Exclusion—2011 to 2020

Throughout the 2000s, the electricity regime had initiated a socio-technical reconfiguration process underpinned by (a) the emergence of a decentralised means for renewable electricity production, driven by (b) a diversified constellation of non-utility actors, and supported by (c) an actor-neutral remuneration scheme of long-term fixed FiTs that "had gained broad social acceptance as it included a wide range of actors—from single households to corporations" [25] (p. 37). By 2012, the distributed generation base of the RES-E market comprised 1.3 million generators, with 47% of RES capacity owned by citizens, 41.5% by small and medium-sized (SME) developers/operators and financial institutions, and 12.5% by utility companies [131,132]. "We had already over 20% renewables. This means the big utilities, their business on electricity production, was reduced by 20%" (Government 1). The market diffusion of RETs became a threat not only to their traditional business model but also to an increasingly obsolete paradigm of centralised electricity supply, a reality they both underestimated and were late to fully recognise (Government 1, IPP 2) [133–135]. Utility companies—who had traditionally held a dominant position in the electricity market—were being relegated to a position of market latecomers in the RES-E market. As noted by [136] (p. 20), "incumbents were about 10 years late [ ... ] in substantially reacting to the government promotion of renewable energies".

At this stage, their closure strategy shifted once again, this time to slow down the pace of expansion of a rapidly growing RES-E market, secure greater control over the business opportunities stemming from it, and in doing so, counteract the market usurpation that RES operators were undertaking in their generation base (Incumbents' responses to Germany's *Energiewende* encompassed a wide range of actions including, e.g., disinvestments in energy supply/distribution, subsidy demands for conventional power units, litigation against nuclear phase-out policies, etc. These and other responses are outside the scope of this study, as they do not directly concern the generation base of the RES-E market nor the feed-in policy support framework upholding its diffusion. See [125,136] for a wider range of incumbent adaptive/reactive strategies beyond those concerning renewables). Although the establishment of renewables subsidiaries/divisions gave them an entry into the RES-E market, their large corporate structures were ill-conceived to profitably operate smaller scale, more modular and decentralised generation units (IPP 1). RWE's 2013 internal reports are exemplary of their operational challenges: "in a low-interest environment, it will not be possible to generate sufficient returns within this subsidised industry. Our cost of capital will not be competitive against funding from private and institutional equity investors" [137] (para. 3–4). Furthering this assessment, IPP 1 noted:

"Many large companies and utilities had too high fixed costs in their structures, they were not able to adapt to a small and medium sized, very dynamic market. So they needed to slow down the growth rate of renewables. Because under a pure FiT system only SMEs were being active and successful in the market. But they are larger, so they knew if they wanted to get into the market they needed a system where their size makes a difference, where they could make use of their assets, which are mainly scale and money. They wanted a system where the smaller companies were not able to compete, so many of them promoted a volume regulation".

To do so, they collectively orchestrated a market framing strategy to steer the public and political debates towards the exorbitant costs of the current FiT-based model for renewables' support. Among other actions, "they paid for this initiative that made huge advertisements in each main station of the public transport system saying that the EEG levy is disastrous" (Figure 5) (Consultant 1). The campaign targeted prohibitive household energy costs to highlight the tension between energy affordability and a competition-exempt, over-subsidised, and cost-deficient RES-E market (Government 1, RES Association 1, Consultant 1) [138]. In effect, the continued increase in the EEG levy for financing renewables' support costs was paired with a 72% rise in household electricity prices between 2000– 2011 [139,140]. Failing to pacify this dichotomy could potentially trigger societal opposition against further RES expansion should the public perceive it as occurring at the expense of increasing energy costs (Consultant 1, E-Association 1). This was taken on as a key point of contention by the market incumbents to argue for the substitution of FiTs with a more market-oriented scheme to support renewables efficiently and within acceptable costs for the German population and the export-oriented industrial base (Government 1, E-Association 1). Both RES auctioning and tradeable renewable certificates were framed as the most competition-enabling, and therefore, the least market distortive policy devices (Utility 1, E-Association 1) [107,141–143].

The market framing strategy proved sufficiently salient for policymakers to reformulate what it characterised as an outdated feed-in policy support framework failing to upkeep with the dynamic growth of a competition-exempt RES-E market (IPP 2, Government 3) [141]. "The government felt huge pressure to change the [support] scheme because of this huge campaign from conventional power producers who saw their business model threatened" (Cooperative 2).



**Figure 5.** "With subsidies, the energy transformation will drown" (**left**) and "Put an end to the horror of electricity prices" (**right**). These print advertisements against over-subsidised renewables triggering rising electricity prices werepart of the 2012 public campaign, "Stop the EEG now!" Source: Reprinted with permission from Ref. [144]. 2012, Initiative Neue Soziale Marktwirtschaft (INSM).

Concurrently, the government started pursuing the market integration of RES-E through demand-driven generation and a market-guiding sales mechanism [145]. This became legally encoded in the 2012 and 2014 EEG amendments, which progressively substituted FiTs for a market-oriented model of renewables' support consisting of direct marketing and sliding feed-in premiums (FiPs) [146,147]. In doing so, the government expanded the end goal of the feed-in policy support framework from securing the investability of renewables as a means to operationalise climate and energy targets, to ensuring their marketability to operationalise increased cost-effectiveness—that is, sufficiently in-

centivising their market responsiveness to demand-based price signals so as to increase operating efficiency and, in doing so, induce cost reductions in RES support (Government 2, IPP 1, Consultant 1) [148]. This served as the first step towards the market integration of RES-E generation—from policy-driven to demand-driven renewables' diffusion (IPP 2).

Premium-based remuneration was shortly followed by the introduction of competitive tendering for allocating RES support. Auctions were first piloted for ground-mounted solar PV during 2015–2016 prior to their full-scale deployment for all RETs in 2017 [149,150]. Their use represented a (second) paradigm shift in the way renewables support had been operationalised thus far (RES Association 2, C-Association 2, Government 3). For the first time since the enactment of Germany's feed-in policy support framework for renewables in 1991, RES-E generation no longer had guaranteed—but conditional—remuneration contingent upon the relative cost-performance of RES developers/operators, who were now impelled to compete for the right to market their RES-E under a premium-based model of support. Auctioning was thus framed as the best-suited policy device to increase the costeffectiveness of RES support: by repurposing remuneration as the object of competition, cost-reductions would be induced, and with it, a more efficient support allocation (Utility 1, Consultant 3) [151]—assuming there were sufficient "competitors" to enact such costreductions (IPP 2). Building upon this assumption, the government further characterised auctions as a policy device conceived to facilitate an actor-diverse participation by large RES operators alongside SMEs, energy cooperatives, and locally-embedded RES developersensuring strong social acceptance for RES expansion [151].

Contrary to such characterisation, the use of auctions favoured larger developers with stronger operational and financial means (i.e., larger project portfolios to spread risk) to place more cost-competitive bids (Consultant 3, IPP 1) [152–154]. On the downside, smaller players with more socially-innovative but risk-exposed business models—such as those instantiated by citizen-driven formats for RES-E generation—became increasingly penalised against the risks associated with having to compete for securing support (C-Association 2, Government 4, Consultant 3) [152,154]. In the first six pilot auctions for solar PV between 2015–2016, only 0.8% of bids were won by energy cooperatives, representing 0.2% of the installed capacity auctioned [155–160].

In light of such bleak outcomes, the 2017 EEG amendment introduced a differentiated set of design features to favour community energy initiatives partaking in onshore wind energy auctions. These included a 24-month extension for project realisation (from 30 to 54 months), halving the bid bond from EUR 30/kW to EUR 15/kW, no prior approval pursuant to the Federal Immission Control Act, and obtaining the highest winning bid [150]. However, a loosely formulated definition of "citizen energy companies" (CECs) enabled professional developers to easily qualify as CECs when in fact their operational structure and financial capital were not reflective of such a legal form (IPP 3, Utility 2, RES Association 2, Government 4) [161,162]. Some of them took advantage of this situation and placed predatory bids knowing that they would not cover their project realisation costs (IPP 2, Utility 2) [163]. This was not so much due to a miscalibration of development costs, but rather strategically pursued to crowd out competitors and retain influence rents. In doing so, they perverted the end-goal of auctions—from a cost-reducing policy device to a tool of market exclusion—to prevent competitors from their continued participation in the RES-E market. As explained by IPP 2:

"These special rules [for CECs] were a joke. All the costs that you would typically have before the auction, those related with the permits to build your installation, they were gone because you didn't have to show you were approved to participate. Basically, these rules allowed you to pay EUR 15/kW to participate in an auction of 1000 MW, so for just EUR 15 million a big player could come in and buy the entire German onshore wind power market. And this is actually what happened. Some companies took advantage of these conditions and placed unrealistically low bids of like EUR 3–4 ct/kWh to outcompete all others, and so they captured almost all the auctioned capacity. So at first sight it looks like an excellent result, you have many of these citizen initiatives participating in the auction and with record-low bid prices. But when you look into it you actually see that many of these citizen initiatives are in fact administered by only a few companies who won almost everything. They abused these special rules to outbid everyone, they basically bought themselves exclusive access to the market and kicked out everyone else. It is obvious that those who were successful in these first rounds [of 2017], they were not bidding to maximise their revenues, they were clearly bidding to maximise their market shares".

In response to this misuse, in 2018 the government abolished the special provisions for CECs and re-introduced stricter rules for all auction participation. This was followed by a progressive increase in support costs due to insufficient competition (i.e., undersubscribed auctions). "There was not enough people to compete, so everyone who participated in the auction got their remuneration guaranteed" (Government 4). "It was basically like a FiT. The goal of cost-efficiency was completely missed, plus all these citizen initiatives are now out of the market" (IPP 2). In effect, insufficient competition was coupled with a drastic drop in CEC participation (Figure 6) (Government 4) [164].



**Figure 6.** Share of capacity volumes submitted by community energy projects (green bars—%) compared with the evolution of average auction strike prices (red dots—EUR ct/kWh) and maximum allowed bidding (ceiling) prices (yellow line—EUR ct/kWh) for all completed onshore wind energy auction rounds conducted in Germany between 2017–2021. Source: Own elaboration. Data from [165].

Ironically, like the days prior to the StrEG's enactment, citizen-led RES developments previously the main driving actor behind Germany's market diffusion of RETs—are yet again becoming displaced towards the margins of a RES-E market this time locked-into the imperatives of allocative efficiency and cost-effectiveness. With the continued use of auctions as the core policy device for allocating RES support, the configurational nature of Germany's RES-E market will likely further reduce its actor diversity, yielding fewer but larger players driving the market diffusion of RETs (IPP 3). As concluded by IPP 1:

"When renewables first took off in the early 2000s, we had hundreds of players, hundreds of developer companies. Then you see a consolidation process because smaller companies were no longer able to take the regulatory risk. This is just the wrong way around. Usually regulation should reduce risk. Instead we have regulation which increases risk and leads to an actor consolidation which is not healthy".

#### 5. Discussion

Taking the German experience as a reference, the case study above illustrates that the emergence and evolution of a country's RES-E market is a highly contested process undertaken by market participants who perceive their dominant position in the marketplace at risk of being usurped by competing participants. Given the policy-driven character of the country's market-mediated diffusion of RETs, such contestation has inevitably revolved around the legitimacy of the feed-in policy support framework for renewables.

First, the case analysis uncovers how the closure strategy of electricity market incumbents has recurrently targeted the "affordability" of the support framework, both through legal means and market framing actions. In both instances, the analysis shows how market incumbents have made a politicised use of support costs to dispute the economic efficiency of FiTs and accentuate their detrimental distributive impacts penalising household electricity consumers via prohibitively expensive electricity prices. The German case thus exemplifies how affordability and costs (along with their distribution) are predominantly viewed as the most influential drivers shaping policy acceptance in an energy transition context [166]. Importantly, the analysis reveals how such drivers are strategically leveraged through the targeted use of public discursive strategies to discredit certain market-shaping policy devices while justifying the need for alternatives [10,65]. This appears as a critical undertaking with potential market-shaping implications, since a core political driver for sustainability-oriented policy formulation revolves around the public's acceptance of its intended purpose and, most importantly, of its distributional impacts across the electoral voting base [167–169]. Steering the socio-political debate through discursive frames that (de)legitimise the value logic of different market-shaping policy devices (e.g., cost-deficient FiTs vs. cost-effective auctions) has therefore been a resourceful means for influencing their assemblage in the context of Germany's policy-driven RES-E market, where socio-economic concerns over policy-induced electricity price increases and energy use affordability have gained increasing relevance as key challenges undermining the political acceptance of FiT-based support [115,167,170,171].

Second, Germany's feed-in policy support framework has not only been the target of market participants' political contestations, but also the vehicle through which they have strategically steered the RES-E market configuration in the pursuit of their respective closure strategies and influence rent-seeking efforts. For instance, early VAAs empowered market incumbents (energy utilities) with a privileged role as market gatekeepers, who strategically repurposed the scheme as a tool for market exclusion to arbitrate market entry for non-utility participants. Consecutively, market newcomers (independent RES operators) found in FiTs the necessary vehicle to secure themselves market entry and initiate a market usurpation process for an incumbent-dominated electricity generation base, which they attempted to consolidate by demanding better support conditions via higher remuneration. Yet in a clear lack of strategic foresight, their influence rent-seeking efforts were pursued at a critical juncture overwhelmingly dominated by rising support costs and electricity prices, fuelling existing political concerns over the affordability of a FiTbased renewables' support framework. This was in turn strategically leveraged by market incumbents through the abovementioned market framing strategy, which contributed to the substitution of administratively-set FiTs for auction-allocated FiPs. It is in this latter shift, with the introduction of competitive tendering via auctioning, where the case analysis elucidates the clearest instantiation of a market co-optation strategy, where a marketshaping policy device originally assembled to increase cost-efficacy in support allocation is strategically repurposed as a tool for market exclusion by a small number of auction participants to grant themselves almost exclusive access to market opportunities derived from FiPs while undermining the ability of competing actors to further partake in the RES-E market.

In sum, Germany's electricity market transition illustrates how the strategic framing of a policy's affordability along with the perversion of its originally-intended purpose become powerful market-shaping resources to drive policy acceptance and influence market change in certain directions. The resulting transition pathways are therefore not neutral but deeply political, in the sense that they tend to benefit traditionally well-organised market actor groups with ample political bargaining power (e.g., utilities, large professional developers/IPPs, energy-intensive industrial consumers) while penalising those ill-organised and with reduced political clout (e.g., citizen energy cooperatives, residential electricity consumers) [8,16,172]. As such, they reproduce (but can also disrupt) asymmetric power relations between market participants unequally endowed to shape the selection environments they operate in [51,69].

On this final note, the present paper attempts to contribute a reflection with respect to the relative endowments of market-shaping policy devices for determining different market actor capabilities to partake in economised relations that would otherwise not unfold naturally without their necessary introduction. The present case study shows how Germany's feed-in policy support framework for renewables has undergone a steady evolution from an environmentally effective policy device elevating the desirability of a RES-E market space in light of its socio-ecological superiority and actor diverse configuration; progressively giving way to a policy device with the overriding concern of competition inducing cost-effectiveness and allocative efficiency. Yet this should not entail a negative cognition of competition as a disruptive market dynamic as long as it effectively contributes to accelerating a socio-technical transition expeditiously enough so as to avoid the detrimental consequences of irreversible climate change. Net socio-ecological welfare outcomes must therefore reflect competition-inducing economic efficiency gains vis-à-vis foregone opportunity costs reflecting the *pace* and *quality* of a market-mediated energy transition, namely:

- Temporal scarcity (pace): pursuing economic efficiency is desirable insofar as it expedites the realisation of climate and energy objectives within an increasingly limited temporal horizon. As such, it must be addressed alongside temporal scarcity if policymakers are to make climate-smart decisions on the most environmentally effective policies to transition national energy systems towards carbon neutrality by mid-century [173,174]. What is the net socio-ecological welfare outcome of delivering a wind-generated kWh at EUR 4 ct if by doing so we forego the deployment of an additional 8 MW of renewable power capacity?
- Distributive impacts (quality): if the use of competition-inducing policies to efficiently allocate RES support is captured by a reduced number of larger market actors at the expense of a broader and more diverse number of smaller competitors, then the pre-condition of actor diversity necessary for the possibility of competition in the marketplace is foregone [175]. As a consequence, the intended economic benefits of competition-inducing efficiency are off-traded by RES support concentration and actor consolidation outcomes. Furthermore, if those market actors endowed with a comparatively stronger social capital (i.e., high community acceptance amongst local residents) [176–178] are also the most penalised by competition-inducing RES support allocation, then the policy objective of social acceptance delegated to auctioning is foregone, and the intended economic distributive effect of an actor-diverse RES-E marketplace is deluded.

In closing, this study calls upon increased reflexivity on: the relative weight that different market-shaping policy devices impose on the quality and pace of market-mediated sustainability transitions; how these are strategically steered by different market participants in their respective attempts to orchestrate certain market (inter)actions furthering their vested interests; and their ensuing impact on the ability of different market actors to participate in a policy-driven yet market-mediated transition towards a more sustainable socio-technical system of energy generation, supply, and use.

#### 6. Conclusions

This study re-visits Germany's transition towards a renewables-based electricity system in order to examine—by means of a longitudinal case study—the politics of policydriven market reconfiguration processes purposively enacted to realise sustainability objectives. The analytical framework outlined beforehand serves to guide the tracing of the market actors' changing "acceptance positions" (e.g., opposition, concern, reluctance, indifference, dismissal, consent, support, etc.) in relation to the country's evolving feed-in policy support framework for orchestrating a market-mediated diffusion of RETs. This is particularly relevant for the German case, whose RES-E market emerged and developed due to the prolonged use of feed-in policy instrumentation, including FiTs, market premia and, more recently, auctioning.

The analysis uncovers how different market actors effectively shape the selection environments they operate in by legally contesting, framing, and misusing the very same policies organising their economised relations and, by extension, steering the actor's configuration of the RES-E market in line with their vested interests. It shows how such efforts are undertaken through legal disputes against specific policy design features, including remuneration levels, eligibility criteria, and pricing schemes. This is coupled with the use of discursive strategies elevating the cost-effectiveness of feed-in policies and challenging their affordability while downplaying their socio-ecological benefits, including social acceptance via citizen participation and climate mitigation via renewables-based energy decarbonisation.

Ultimately, the prevailing use of market-shaping practices, if left unchecked or unaccounted for in the design of RES policy support frameworks, might eventually skew RES-E market participation in favour of a reduced number of large market players with greater financial and operational endowments to operate in selection environments with an overriding concern for economic efficiency. Policymaking efforts should therefore assess the net socio-ecological welfare outcome resulting from the relative benefits of economic efficiency along with the environmental effectiveness and distributive impacts of RES policy support instrumentation, reflecting a more balanced application of efficiency, pace, and quality considerations for securing social acceptance for a policy-driven yet market-mediated energy transition.

This study has limitations that pave the way for future research. This is particularly the case with respect to the single case study approach adopted for this analytical exercise. While Germany stands as a representative national experience informing the ensuing policy choices of other EU Member States, its experience can only be extrapolated to other national jurisdictions to a certain extent, particularly with regards to non-EU countries, such as the United Kingdom, Norway, or Switzerland. The generalisability of the reported findings is therefore limited in scope and reach. In this respect, future research on the politics of market change towards sustainability could adopt a cross-country, comparative case study research design. This would enable the identification of similarities/differences in the pervasiveness and diversity of market-shaping practices beyond national jurisdictions, along with the extent by which international firms operating in different RES-E markets replicate/expand market-shaping strategies across national borders. This would in turn enable an assessment of the degree of institutionalisation of such contentious strategies in the corporate cultures of different RES-E market participants, as well as support more generalizable findings substantiating a broader and more diverse set of policy implications.

**Funding:** This research is funded by the European Union's Horizon 2020 Research and Innovation Programme for funding the project MISTRAL under the Marie Skłodowska-Curie Actions [Grant agreement N°. 813837]. The views and opinions expressed in this publication are the sole responsibility of the author and do not necessarily reflect the views of the European Commission. Research design, data collection and analysis remain within the sole responsibility of the author. All remaining errors are the author's own.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

**Data Availability Statement:** Relevant data informing this study can be obtained via the following resources. Legal documentation and parliamentary proceedings concerning European legislation/regulation can be found at the EU's online legal archive (https://eur-lex.europa.eu/homepage.html, accessed on 17 March 2020) and concerning German legislation/regulation and parliamentary proceedings at the German Federal Gazette's online legal repository (https://www.bgbl.de/xaver/bgbl/start.xav, accessed on 20 March 2020). Data concerning statistical datasets on

support costs, electricity prices, and consumption profiles can be accessed on Eurostat's energyrelated database (https://ec.europa.eu/eurostat/web/main/data/database, accessed on 2 February 2020) as well as on BMWi's information portal for renewable energies (https://www.erneuerbareenergien.de/EE/Navigation/DE/Home/home.html, accessed on 2 February 2020), the German Federal Network Agency's website (https://www.bundesnetzagentur.de/, accessed on 5 February 2020), and on the information platform of the four German transmission system operators (https://www.netztransparenz.de/, accessed on 3 February 2020). Concerning the number and structure of Germany's electricity market, data can be retrieved from the German energy market data register online portal (MaStR—https://www.marktstammdatenregister.de/MaStR, accessed on 11 April 2020) and from the German Federal Network Agency's information platform for the German electricity market (SMARD—https://www.smard.de/, accessed on 11 April 2020).

Acknowledgments: The author would like to thank the constructive comments provided by IMES colleagues Hannes Sonnsjö, Jamil Kahn, and Alexandra Nikoleris, who jointly contributed to improve and strengthen the original manuscript. The author is also grateful for the valuable feedback provided by two anonymous reviewers during the formal review process.

**Conflicts of Interest:** The author declares no conflict of interest. The funders had no role in the design of the study; in the collection, analyses, or interpretation of data; in the writing of the manuscript, or in the decision to publish the results.

#### References

- 1. Avelino, F.; Wittmayer, J.M. Shifting Power Relations in Sustainability Transitions: A Multi-Actor Perspective. J. Environ. Policy Plan. 2016, 18, 628–649. [CrossRef]
- Farla, J.; Markard, J.; Raven, R.; Coenen, L. Sustainability Transitions in the Making: A Closer Look at Actors, Strategies and Resources. *Technol. Forecast. Soc. Chang.* 2012, 79, 991–998. [CrossRef]
- Wolsink, M. Social Acceptance Revisited: Gaps, Questionable Trends, and an Auspicious Perspective. *Energy Res. Soc. Sci.* 2018, 46, 287–295. [CrossRef]
- 4. European Commission. Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the regions. In 'Fit for 55': Delivering the EU's 2030 Climate Target on the Way to Climate Neutrality (COM/2021/550 Fi); European Commission: Brussels, Belgium, 2021.
- 5. International Energy Agency. Renewables 2021—Analysis and Forecasts to 2026; International Energy Agency: Paris, France, 2021.
- 6. Cointe, B.; Nadaï, A. Feed-in Tariffs in the European Union: Renewable Energy Policy, the Internal Electricity Market and Economic Expertise; Cointe, B., Nadaï, A., Eds.; Palgrave Macmillia: New York, NY, USA, 2018; ISBN 9783319763200.
- 7. Strunz, S. The German Energy Transition as a Regime Shift. *Ecol. Econ.* **2014**, *100*, 150–158. [CrossRef]
- Gawel, E.; Strunz, S.; Lehmann, P. Support Policies for Renewables: Instrument Choice and Instrument Change from a Public Choice Perspective. In *The Political Economy of Clean Energy Transitions*; Arent, D., Arndt, C., Miller, M., Tarp, F., Zinaman, O., Eds.; Oxford University Press: Oxford, UK, 2017; pp. 80–102. ISBN 978-0-19-880224-2.
- 9. Kungl, G. The Incumbent German Power Companies in a Changing Environment: A Comparison of E.ON, RWE, EnBW and Vattenfall from 1998 to 2013; Stuttgart: Stuttgart, Germany, 2014.
- 10. Strunz, S.; Gawel, E.; Lehmann, P. The Political Economy of Renewable Energy Policies in Germany and the EU. *Util. Policy* **2016**, 42, 33–41. [CrossRef]
- 11. Fuchs, G. Legitimacy and Field Development: Electricity Transition(s) in Germany. Glob. Transit. 2019, 1, 141–147. [CrossRef]
- 12. Stenzel, T.; Frenzel, A. Regulating Technological Change—The Strategic Reactions of Utility Companies towards Subsidy Policies in the German, Spanish and UK Electricity Markets. *Energy Policy* **2008**, *36*, 2645–2657. [CrossRef]
- Köhler, J.; Geels, F.W.; Kern, F.; Markard, J.; Onsongo, E.; Wieczorek, A.; Alkemade, F.; Avelino, F.; Bergek, A.; Boons, F.; et al. An Agenda for Sustainability Transitions Research: State of the Art and Future Directions. *Environ. Innov. Soc. Transit.* 2019, 31, 1–32. [CrossRef]
- 14. Avelino, F.; Grin, J.; Pel, B.; Jhagroe, S. The Politics of Sustainability Transitions. J. Environ. Policy Plan. 2016, 18, 557–567. [CrossRef]
- 15. Kemp, R.; Weaver, P.M.; Strasser, T.; Backhaus, J.; Golland, A. *Socio-Economic Transformations: Insights for Sustainability*; Publications Office of the European Union: Luxembourg, 2018.
- 16. Van Oers, L.; Feola, G.; Moors, E.; Runhaar, H. The Politics of Deliberate Destabilisation for Sustainability Transitions. *Environ. Innov. Soc. Transit.* **2021**, *40*, 159–171. [CrossRef]
- 17. Hausknost, D.; Haas, W. The Politics of Selection: Towards a Transformative Model of Environmental Innovation. *Sustainability* **2019**, *11*, 506. [CrossRef]
- Turnheim, B.; Asquith, M.; Geels, F.W. Making Sustainability Transitions Research Policy-Relevant: Challenges at the Science-Policy Interface. *Environ. Innov. Soc. Transit.* 2020, 34, 116–120. [CrossRef]

- Ratinen, M.; Lund, P.D. Democracy and Transitions: European Experiences of Policy Inclusiveness and Changes in the Electricity Industry. In *The Palgrave Handbook of the International Political Economy of Energy*; Van de Graaf, T., Sovacool, B.K., Ghosh, A., Kern, F., Klare, M.T., Eds.; Palgrave Macmillan: London, UK, 2016; pp. 341–362. ISBN 978-1-137-55631-8.
- Kern, F.; Markard, J. Analysing Energy Transitions: Combining Insights from Transition Studies and International Political Economy. In *The Palgrave Handbook of the International Political Economy of Energy;* Van de Graaf, T., Sovacool, B.K., Ghosh, A., Kern, F., Klare, M.T., Eds.; Palgrave Macmillan: London, UK, 2016; pp. 291–318. ISBN 978-1-137-55631-8.
- 21. Boon, W.P.C.; Edler, J.; Robinson, D.K.R. Market Formation in the Context of Transitions: A Comment on the Transitions Agenda. *Environ. Innov. Soc. Transit.* 2020, 34, 346–347. [CrossRef]
- Cointe, B. The Emergence of Photovoltaics in France in the Light of Feed-in Tariffs: Exploring the Markets and Politics of a Modular Technology. Ph.D. Thesis, Centre International de Recherche sur l'Eenvironnement et le Développement (CIRED), Paris, France, 2014.
- 23. Jacobsson, S.; Lauber, V. The Politics and Policy of Energy System Transformation—Explaining the German Diffusion of Renewable Energy Technology. *Energy Policy* 2006, 34, 256–276. [CrossRef]
- 24. Lauber, V.; Mez, L. Renewable Electricity Policy in Germany, 1974 to 2005. Bull. Sci. Technol. Soc. 2006, 26, 105–120. [CrossRef]
- Leiren, M.D.; Reimer, I. Historical Institutionalist Perspective on the Shift from Feed-in Tariffs towards Auctioning in German Renewable Energy Policy. *Energy Res. Soc. Sci.* 2018, 43, 33–40. [CrossRef]
- Stefes, C.H. Bypassing Germany's Reformstau: The Remarkable Rise of Renewable Energy. Ger. Polit. 2010, 19, 148–163. [CrossRef]
- 27. Sturm, C. How the Energiewende Came to Be. In *Lecture Notes in Energy*; Springer: Berlin/Heidelberg, Germany, 2020; Volume 75, pp. 67–121.
- 28. Diaz Ruiz, C.; Makkar, M. Market Bifurcations in Board Sports: How Consumers Shape Markets through Boundary Work. *J. Bus. Res.* **2021**, *122*, 38–50. [CrossRef]
- Silvast, A. Energy, Economics, and Performativity: Reviewing Theoretical Advances in Social Studies of Markets and Energy. Energy Res. Soc. Sci. 2017, 34, 4–12. [CrossRef]
- Kjellberg, H.; Azimont, F.; Reid, E. Market Innovation Processes: Balancing Stability and Change. Ind. Mark. Manag. 2015, 44, 4–12. [CrossRef]
- Nenonen, S.; Kjellberg, H.; Pels, J.; Cheung, L.; Lindeman, S.; Mele, C.; Sajtos, L.; Storbacka, K. A New Perspective on Market Dynamics: Market Plasticity and the Stability–Fluidity Dialectics. *Mark. Theory* 2014, 14, 269–289. [CrossRef]
- Ottosson, M.; Magnusson, T.; Andersson, H. Shaping Sustainable Markets—A Conceptual Framework Illustrated by the Case of Biogas in Sweden. *Environ. Innov. Soc. Transit.* 2020, 36, 303–320. [CrossRef]
- Çalişkan, K.; Callon, M. Economization, Part 2: A Research Programme for the Study of Markets. *Econ. Soc.* 2010, 39, 1–32. [CrossRef]
- 34. Muniesa, F.; Millo, Y.; Callon, M. An Introduction to Market Devices. Sociol. Rev. 2007, 55, 1–12. [CrossRef]
- 35. Nenonen, S.; Storbacka, K. Market-Shaping: Navigating Multiple Theoretical Perspectives. AMS Rev. 2021, 11, 336–353. [CrossRef]
- Storbacka, K.; Nenonen, S. Scripting Markets: From Value Propositions to Market Propositions. Ind. Mark. Manag. 2011, 40, 255–266. [CrossRef]
- Ulkuniemi, P.; Araujo, L.; Tähtinen, J. Purchasing as Market-Shaping: The Case of Component-Based Software Engineering. *Ind. Mark. Manag.* 2015, 44, 54–62. [CrossRef]
- 38. Overdevest, C. Towards a More Pragmatic Sociology of Markets. Theory Soc. 2011, 40, 533–552. [CrossRef]
- 39. Onyas, W.I.; Ryan, A. Agencing Markets: Actualizing Ongoing Market Innovation. Ind. Mark. Manag. 2015, 44, 13–21. [CrossRef]
- 40. Valor, C.; Lind, L.; Cossent, R.; Escudero, C. Understanding the Limits to Forming Policy-Driven Markets in the Electricity Sector. *Environ. Innov. Soc. Transit.* **2021**, *40*, 645–662. [CrossRef]
- Lesser, J.A.; Su, X. Design of an Economically Efficient Feed-in Tariff Structure for Renewable Energy Development. *Energy Policy* 2008, 36, 981–990. [CrossRef]
- 42. King, B.G.; Pearce, N.A. The Contentiousness of Markets: Politics, Social Movements, and Institutional Change in Markets. *Annu. Rev. Sociol.* **2010**, *36*, 249–267. [CrossRef]
- 43. Quitzow, R. Dynamics of a Policy-Driven Market: The Co-Evolution of Technological Innovation Systems for Solar Photovoltaics in China and Germany. *Environ. Innov. Soc. Transit.* 2015, *17*, 126–148. [CrossRef]
- 44. Quitzow, R.; Walz, R.; Köhler, J.; Rennings, K. The Concept of 'Lead Markets' Revisited: Contribution to Environmental Innovation Theory. *Environ. Innov. Soc. Transit.* 2014, 10, 4–19. [CrossRef]
- 45. Fligstein, N. Markets as Politics: A Political-Cultural Approach to Market Institutions. *Am. Sociol. Rev.* **1996**, *61*, 656–673. [CrossRef]
- Holm, P.; Nielsen, K.N. Framing Fish, Making Markets: The Construction of Individual Transferable Quotas (ITQs). Sociol. Rev. 2007, 55, 173–195. [CrossRef]
- Mathews, J.A. How Carbon Credits Could Drive the Emergence of Renewable Energies. *Energy Policy* 2008, 36, 3633–3639. [CrossRef]
- Bumpus, A.G. The Matter of Carbon: Understanding the Materiality of TCO2e in Carbon Offsets. *Antipode* 2011, 43, 612–638. [CrossRef]

- 49. Laurent, B. The Politics of European Agencements: Constructing a Market of Sustainable Biofuels. *Environ. Polit.* 2015, 24, 138–155. [CrossRef]
- Grandclément, C.; Nadaï, A. Transitioning Through Markets. In *Energy Transitions: A Socio-Technical Inquiry*; Labussière, O., Nadaï, A., Eds.; Springer International Publishing: Cham, Switzerland, 2018; pp. 101–142. ISBN 978-3-319-77025-3.
- 51. White, G. Towards a Political Analysis of Markets. *IDS Bull.* **2016**, 47, 45–57. [CrossRef]
- 52. Harriss-White, B. Market Politics and Climate Change. Development 2008, 51, 350–358. [CrossRef]
- 53. Hess, D.J. Sustainability Transitions: A Political Coalition Perspective. Res. Policy 2014, 43, 278–283. [CrossRef]
- 54. Roithmayr, D. Barriers to Entry: A Market Lock-in Model of Discrimination. Va. Law Rev. 2000, 86, 727–799. [CrossRef]
- 55. Safarzyńska, K.; van den Bergh, J.C.J.M. Demand-Supply Coevolution with Multiple Increasing Returns: Policy Analysis for Unlocking and System Transitions. *Technol. Forecast. Soc. Chang.* **2010**, *77*, 297–317. [CrossRef]
- 56. Unruh, G.C. Understanding Carbon Lock-In. Energy Policy 2000, 28, 817–830. [CrossRef]
- 57. Roberts, C.; Geels, F.W. Conditions for Politically Accelerated Transitions: Historical Institutionalism, the Multi-Level Perspective, and Two Historical Case Studies in Transport and Agriculture. *Technol. Forecast. Soc. Chang.* **2019**, *140*, 221–240. [CrossRef]
- Astuti, S.P.; Day, R.; Emery, S.B. A Successful Fuel Transition? Regulatory Instruments, Markets, and Social Acceptance in the Adoption of Modern LPG Cooking Devices in Indonesia. *Energy Res. Soc. Sci.* 2019, 58, 101248. [CrossRef]
- Kim, S.E.; Urpelainen, J.; Yang, J. Electric Utilities and American Climate Policy: Lobbying by Expected Winners and Losers. J. Public Policy 2016, 36, 251–275. [CrossRef]
- 60. Kumar, R. *Rethinking Revolutions: Soyabean, Choupals, and the Changing Countryside in Central India;* Oxford University Press: Delhi, India, 2016; ISBN 9780199465330.
- 61. Cawson, A.; Morgan, K.; Webber, D.; Holmes, P.; Stevens, A. *Hostile Brothers: Competition and Closure in the European Electronics Industry*; Clarendon Press: Oxford, UK, 1990; ISBN 0198275684.
- 62. Klemperer, P. How (Not) to Run Auctions: The European 3G Telecom Auctions. Eur. Econ. Rev. 2002, 46, 829–845. [CrossRef]
- 63. Hughes, L.; Meckling, J. The Politics of Renewable Energy Trade: The US-China Solar Dispute. *Energy Policy* **2017**, *105*, 256–262. [CrossRef]
- 64. Voituriez, T.; Wang, X. Real Challenges behind the EU–China PV Trade Dispute Settlement. *Clim. Policy* **2015**, *15*, 670–677. [CrossRef]
- 65. Rinscheid, A. Business Power in Noisy Politics: An Exploration Based on Discourse Network Analysis and Survey Data. *Polit. Gov.* **2020**, *8*, 286–297. [CrossRef]
- 66. Cawson, A. The Political Analysis of Economic Markets. *IDS Bull.* 1993, 24, 63–67. [CrossRef]
- 67. Weber, M. Economy and Society: An Outline of Interpretive Sociology; Roth, G., Wittich, C., Eds.; University of California Press: Los Angeles, CA, USA, 1978; ISBN 0520035003.
- 68. Ahuja, G.; Yayavaram, S. Perspective—Explaining Influence Rents: The Case for an Institutions-Based View of Strategy. *Organ. Sci.* **2011**, *22*, 1631–1652. [CrossRef]
- 69. Sinha, S. The Politics of Markets: Farmer–Trader Relations under Neoliberalism in Punjab, India. J. Agrar. Chang. 2020, 20, 255–269. [CrossRef]
- Kemp, R.P.M.; Rip, A.; Schot, J. Constructing Transition Paths Through the Management of Niches. In *Path Dependence and Creation*; Garud, R., Karnoe, P., Eds.; Lawrence Erlbaum: Mahwa, NJ, USA; London, UK, 2001; pp. 269–299. ISBN 0-8058-32726.
- Smith, A.; Raven, R. What Is Protective Space? Reconsidering Niches in Transitions to Sustainability. *Res. Policy* 2012, 41, 1025–1036. [CrossRef]
- 72. Geels, F.W. Socio-Technical Transitions to Sustainability. In *Perspectives on Transitions to Sustainability*; Asquith, M., Lung, T., Eds.; European Environment Agency: Copenhagen, Denmark, 2017; pp. 45–69. ISBN 978-92-9213-939-1.
- Baker, J.J.; Storbacka, K.; Brodie, R.J. Markets Changing, Changing Markets: Institutional Work as Market Shaping. *Mark. Theory* 2019, 19, 301–328. [CrossRef]
- 74. Clausen, N.-E.; Rudolph, D.; Kirkegaard, J.; Vammen Larsen, S. Where to Put Wind Farms? Challenges Related to Planning, EIA and Social Acceptance. In *DTU International Energy Report 2021: Perspectives on Wind Energy*; Holst Jørgensen, B., Hauge Madsen, P., Giebel, G., Martí, I., Thomsen, K., Eds.; DTU Wind Energy: Roskilde, Denmark, 2021; pp. 44–53. ISBN 978-87-87335-00-3.
- 75. Kindström, D.; Ottosson, M.; Carlborg, P. Unraveling Firm-Level Activities for Shaping Markets. *Ind. Mark. Manag.* **2018**, *68*, 36–45. [CrossRef]
- Nenonen, S.; Storbacka, K.; Frethey-Bentham, C. Is Your Industrial Marketing Work Working? Developing a Composite Index of Market Change. Ind. Mark. Manag. 2019, 80, 251–265. [CrossRef]
- 77. Batel, S.; Rudolph, D. Contributions, Tensions and Future Avenues of a Critical Approach to the Social Acceptance of Renewable Energy Infrastructures. In A Critical Approach to the Social Acceptance of Renewable Energy Infrastructures: Going beyond Green Growth and Sustainability; Batel, S., Rudolph, D., Eds.; Springer International Publishing: Cham, Switzerland, 2021; pp. 237–257. ISBN 978-3-030-73699-6.
- George, A.L.; Bennett, A. Case Studies and Theory Development in the Social Sciences; MIT Press: Cambridge, MA, USA, 2005; ISBN 9780262072571.
- 79. Pettigrew, A.M. Longitudinal Field Research on Change: Theory and Practice. Organ. Sci. 1990, 1, 267–292. [CrossRef]

- Ragin, C.C. Turning the Tables: How Case-Oriented Research Challenges. In *Rethinking Social Inquiry: Diverse Tools, Shared Standards*; Brady, H.E., Collier, D., Eds.; Roman and Littlefield Publishers Inc.: Lanham, MD, USA, 2004; Volume 123, ISBN 9781442203440.
- 81. Gebhardt, G.F.; Carpenter, G.S.; Sherry, J.F. Creating a Market Orientation: A Longitudinal, Multifirm, Grounded Analysis of Cultural Transformation. *J. Mark.* 2006, *70*, 37–55. [CrossRef]
- 82. Giesler, M. Conflict and Compromise: Drama in Marketplace Evolution. J. Consum. Res. 2008, 34, 739–753. [CrossRef]
- 83. Nenonen, S.; Fehrer, J.; Brodie, R.J. Editorial: JBR Special Issue on Market Shaping and Innovation. J. Bus. Res. 2021, 124, 236–239. [CrossRef]
- 84. Auer, S. Merkel's Germany and the European Union: Between Emergency and the Rule of Rules. *Gov. Oppos.* **2021**, *56*, 1–19. [CrossRef]
- 85. Morgan, W. New German leader proposes a 'climate club' of leading economies that would punish free riders like Australia. In *The Conversation*; Melbourne, Australia; 2022; p. 5. Available online: https://theconversation.com/new-german-leader-proposesa-climate-club-of-leading-economies-that-would-punish-free-riders-like-australia-175842 (accessed on 3 February 2022).
- 86. Waldheim, S. Bird & Bird. Available online: https://www.twobirds.com/en/insights/2021/germany/german-recovery-and-resilience-plan-an-exemplary-role-in-europe (accessed on 14 February 2022).
- Vögele, S.; Ball, C. Germany: Frontrunner in Europe with Respect to Energy System Transition? In *The European Dimension of Germany's Energy Transition: Opportunities and Conflicts*; Gawel, E., Strunz, S., Lehmann, P., Purkus, A., Eds.; Springer International Publishing: Cham, Switzerland, 2019; pp. 241–252. ISBN 978-3-030-03374-3.
- Gerring, J. Case Selection for Case-Study Analysis: Qualitative and Quantitative Techniques. In *The Oxford Handbook of Political Methodology*; Box-Steffensmeier, J.M., Brady, H.E., Collier, D., Eds.; Oxford University Press: Oxford, UK, 2008; ISBN 9780191577307.
- 89. Flyvbjerg, B. Five Misunderstandings About Case-Study Research. Qual. Ing. 2006, 12, 219–245. [CrossRef]
- 90. Flyvbjerg, B. Case Study. In *The SAGE Handbook of Qualitative Research;* Denzin, N.K., Lincoln, Y.S., Eds.; SAGE Publications: Thousand Oaks, CA, USA, 2011; ISBN 978-1412974172.
- 91. Parker, C.; Scott, S.; Geddes, A. Snowball Sampling. In *SAGE Research Methods Foundations*; Atkinson, P., Delamont, S., Cernat, A., Sakshaug, J.W., Williams, R.A., Eds.; University of Gloucestershire: Cheltenham, UK, 2019; ISBN 9781529747614.
- 92. Bruns, E.; Ohlhorst, D.; Wenzel, B.; Köppel, J. *Renewable Energies in Germany's Electricity Market*; Bruns, E., Ohlhorst, D., Wenzel, B., Köppel, J., Eds.; Springer: Dordrecht, The Netherlands, 2011; ISBN 978-90-481-9905-1.
- 93. Bechberger, M. Forschungsstelle für Umweltpolitik—Das Erneuerbare-Energien-Gesetz (EEG): Eine Analyse des Politikfor-Mulierungsprozesses; Freie Universität Berlin: Berlin, Germany, 2000.
- 94. Berchem, A. Das Unterschätzte Gesetz. Zeit Online 2006, 25, 2006.
- 95. Oelker, J. Windgesichter—Aufbruch der Windenergie in Deutschland, 1st ed.; Sonnenbuch: Dresden, Germany, 2005; ISBN 3-9809956-2-3.
- 96. Kühn, I. International Market for Green Electricity. Overview on German Policy and Opinions among German Market Actors. Renewable Electricity in Germany History of Renewable Sources of Energy in Germany. In *Centre for European Economic Research* (ZEW); Environmental and Resource Economics Department: Mannheim, Germany, 1999; pp. 1–14.
- 97. BMWi. Time Series for the Development of Renewable Energy Sources in Germany; BMWi: Berlin, Germany, 2021.
- Stefes, C.H. Energiewende: Critical Junctures and Path Dependencies Since 1990. In *Rapide Politikwechsel in der Bundesrepublik: Theoretischer Rahmen und Empirische Befunde*; Rüb, F.W., Ed.; Nomos Verlagsgesellschaft mbH & Co. KG: Baden-Baden, Germany, 2014; pp. 48–71. ISBN 978-3-8452-5008-3.
- 99. Durstewitz, M.; Ensslin, C.; Hahn, B.; Hoppe-Klipper, M. The Scientific Measurement and Evaluation Programme (WMEP). Annual Evaluation 1996; Wissenschaftliches Mess- und Evaluierungsprogramm (WMEP) zum Breitentest '250 MW Wind'; Jahresauswertung; ETDEWEB: Oak Ridge, TN, USA, 1996.
- 100. Hoppe-Kilpper, M. Perspektiven der Windenenergienutzung in Deutschland: Zkukünftige Anforderungen an Forschung, Entwicklung und Markterschließung; Zentralbibliothek: Jülich, Germany, 2004.
- 101. Deutscher Bundestag. *Gesetz über die Einspeisung von Strom aus Erneuerbaren Energien in das Öffentliche Netz (Stromeinspeisungsgesetz);* Deutscher Bundestag: Bonn, Germany, 1990; pp. 2633–2634.
- 102. Jahn, D. Nuclear Power Energy Policy and New Politics in Sweden and Germany. Environ. Polit. 1992, 1, 383–417. [CrossRef]
- 103. European Commission. Commission Closes Investigation into Grid Feed-In Law (Germany); European Commission: Ispra, Italy, 2001; p. 2.
- 104. Deutscher Bundestag. Gesetzentwurf der Fraktionen der CDU/CSU und FDP—Entwurf Eines Gesetzes über die Einspeisung von Strom Aus Erneuerbaren Energien in das Öffentliche Netz (Stromeinspeisungsgesetz); Deutscher Bundestag: Berlin, Germany, 1990; pp. 11–7816.
- 105. Nuclear Energy Information Group. Anyone Who Asks Critical Questions Is Far from Being an Opponent of Nuclear Power. Available online: https://www.enklip.de/besondere-quellen.html (accessed on 18 January 2022).
- 106. Pieprzyk, B.; Hilje, P.R. Renewable Energy—Predictions and Reality: Comparison of Forecasts and Scenarios with the Actual Development of Renewable Energy Sources Germany—Europe—The World; Agentur für Erneuerbare Energien: Berlin, Germany, 2009.
- 107. Hirschl, B. Erneuerbare Energien-Politik im Strombereich in Deutschland. In Erneuerbare Energien-Politik: Eine Multi-Level Policy-Analyse mit Fokus auf den Deutschen Strommarkt; VS Verlag für Sozialwissenschaften: Wiesbaden, Germany, 2008; pp. 59–196. ISBN 978-3-531-90890-8.

- 108. Jacobs, D. Renewable Energy Policy Convergence in the EU: The Evolution of Feed-in Tariffs in Germany, Spain and France, 1st ed.; Routledge Taylor & Francis Group: London, UK, 2012; ISBN 9781315605340.
- 109. Deutscher Bundestag. Plenarprotokoll 11/224—Stenographischer Bericht 224; Sitzung: Berlin, Germany, 1990.
- 110. Molly, J.P. Status der Windenenergienutzung in Deutschland; Deutsches Windenergie Institut: Berlin, Germany, 2006.
- Byzio, A.; Heine, H.; Mautz, R.; Rosenbaum, W. Zwischen Solidarhandeln und Marktorientierung. Ökologische Innovation in Selbstorganisierten Projekten—Autofreies Wohnen, Car Sharing und Windenergienutzung; Sozologisches Forschungsinst: Göttingen, Germany, 2002.
- 112. Deustcher Bundestag. Gesetz zur Neuregelung des Energiewirtschaftsrechts; Deustcher Bundestag: Bonn, Germany, 1998; pp. 1–7.
- 113. Lauber, V.; Mez, L. Three Decades of Renewable Electricity Policies in Germany. Energy Environ. 2004, 15, 599–623. [CrossRef]
- 114. Advocate General Jacobs. Opinion of Mr Advocate General Jacobs Delivered on 26 October 2000. PreussenElektra AG v Schhleswag AG, in the Presence of Windpark Reußenköge III GmbH and Land Schleswig-Holstein. Reference for a Preliminary Ruling: Landgericht Kiel—Germany. Electricit; Court of Justice of the European Union: Luxembourg, 2000; pp. 1–56.
- 115. Lauber, V.; Jacobsson, S. The Politics and Economics of Constructing, Contesting and Restricting Socio-Political Space for Renewables—The German Renewable Energy Act. *Environ. Innov. Soc. Transit.* **2016**, *18*, 147–163. [CrossRef]
- 116. BMU. Act on Granting Priority to Renewable Energy Sources (Renewable Energy Sources Act); Deutscher Bundestag: Berlin, Germany, 2000; pp. 1–29.
- 117. European Court of Justice. Judgment of the Court of 13 March 2001. PreussenElektra AG v Schhleswag AG, in the Presence of Windpark Reußenköge III GmbH and Land Schleswig-Holstein. Reference for a Preliminary Ruling: Landgericht Kiel—Germany. Electricity— Renewable Sources of Ene; European Court of Justice: Luxembourg, 2001.
- 118. European Commission. Commission Staff Working Document—The Support of Electricity from Renewable Energy Sources— Accompanying Document to the Proposal for a Directive of the European Parliament and of the Council on the Promotion of the Use of Energy from Renewable Sources; European Commission: Brussels, Belgium, 2008.
- 119. Matthes, F.C.; Grashof, K.; Gores, S. Power Generation Market Concentration in Europe 1996–2005. An Empirical Analysis; Oko Institute: Berlin, Germany, 2007.
- 120. Bundeskartellamt. Sector Inquiry into Electricity Generation and Wholesale Markets; Bundeskartellamt: Bonn, Germany, 2011.
- 121. Trend:Research. Marktakteure Erneuerbare Energien—Anlagen in der Stromerzeugung; Trend:Research: Bremen, Germany, 2011.
- 122. Fell, H.-J. Feed-in Tariff for Renewable Energies: An Effective Stimulus Package without New Public Borrowing; Significance of Renewable Energies in the Current Economic Crisis; ResearchGate: Berlin, Germany, 2009; Volume 90.
- 123. Debor, S. Multiplying Mighty Davids? The Influence of Energy Cooperatives on Germany's Energy Transition; Contributions to Economics; Springer International Publishing: Cham, Switzerland, 2018; ISBN 978-3-319-77627-9.
- Wierling, A.; Schwanitz, V.J.; Zeiß, J.P.; Bout, C.; Candelise, C.; Gilcrease, W.; Gregg, J.S. Statistical Evidence on the Role of Energy Cooperatives for the Energy Transition in European Countries. *Sustainability* 2018, 10, 3339. [CrossRef]
- 125. Gründinger, W. Drivers of Energy Transition How Interest Groups Influenced Energy Politics in Germany, 1st ed.; Springer Fachmeiden Wiesbaden: Berlin/Heidelberg, Germany, 2017; ISBN 978-3-658-17691-4.
- 126. Gawel, E.; Lehmann, P.; Korte, K.; Strunz, S.; Bovet, J.; Köck, W.; Massier, P.; Löschel, A.; Schober, D.; Ohlhorst, D.; et al. The Future of the Energy Transition in Germany. *Energy Sustain. Soc.* **2014**, *4*, 15. [CrossRef]
- 127. BMWi. EEG in Zahlen: Vergütungen, Differenzkosten und EEG-Umlage 2000 Bis 2021; BMWi: Berlin, Germany, 2020.
- 128. IRENA. Renewable Power Generation Costs in 2019; IRENA: Abu Dhabi, United Arab Emirates, 2019.
- 129. Erdmann, G.; Fischedick, M.; von Hirschhausen, C.; Hohmeyer, O.; Jochem, E.; Kemfert, C.; Matthes, F.; Pehnt, M.; Ragwitz, M.; Schmid, J. Dringender Appell Zur Rettung Des Erneuerbare-Energien-Gesetzes Seitens Deutscher Energiewissenschaftler. *DIW Berl. Pressemitt. Vom* 2010, 15, 2010.
- 130. Vorholz, F. Rösler Macht Krawall der Wirtschaftsminister Kämpft Gegen Solar-Subventionen. Wirklich Intelligent Ist Das Nicht; Zeit Online: Berlin, Germany, 2012.
- 131. Leuphana Universität. Definition und Marktanalyse von Bürgerenergie in Deutschland; Trend Research; Leuphana Universität: Bremen/Lünenburg, Germany, 2013.
- 132. BMUV. Development of Renewable Energy Sources in Germany 2011—Graphics and Tables Version; BMUV: Berlin/Bonn, Germany, 2012.
- 133. Richter, M. German Utilities and Distributed PV: How to Overcome Barriers to Business Model Innovation. *Renew. Energy* 2013, 55, 456–466. [CrossRef]
- 134. Steitz, C.; Käckenhoff, T. Germany's RWE Sees No Swift Return to Profit Growth; Reuters: London, UK, 2015.
- 135. Schlandt, J. Can Germany's Energy Giants Change Their DNA? Clean Energy Wire: Berlin, Germany, 2015.
- 136. Kungl, G. Stewards or Sticklers for Change? Incumbent Energy Providers and the Politics of the German Energy Transition. *Energy Res. Soc. Sci.* **2015**, *8*, 13–23. [CrossRef]
- 137. Beckman, K. Exclusive: RWE Sheds Old Business Model, Embraces Transition; Energy Post: Berlin, Germany, 2013.
- 138. Initiative Neue Soziale Marktwirtschaft Energy Transition. Available online: https://www.insm.de/insm/kampagne/ energiewende (accessed on 11 August 2021).
- 139. Eurostat. Electricity Prices for Domestic Consumers—Bi-Annual Data (until 2007); Eurostat: Luxembourg, 2021.
- 140. Eurostat. Electricity Prices for Household Consumers—Bi-Annual Data (from 2007 Onwards); Eurostat: Luxembourg, 2022.
- 141. Frondel, M.; Schmidt, C.M.; Moore, N. Marktwirtschaftliche Energiewende: Ein Wettbewerbsrahmen für die Stromversorgung mit Alternativen Technologien; Springer Vieweg: Essen, Germany, 2012.

- 142. European Commission. Commission Staff Working Document Review of European and National Financing of Renewable Energy in Accordance with Article 23(7) of Directive 2009/28/EC Accompanying Document to the Communication from the Commission to the European Parliament and the Coun; European Commission: Brussels, Belgium, 2011.
- 143. Monopolkommission. Energie 2013: Wettbewerb in Zeiten der Energiewende; Monopolkommission: Bonn, Germany, 2013.
- 144. Initiative Neue Soziale Marktwirtschaft Save the Energy Transition. Available online: https://www.insm.de/insm/kampagne/ energiewende/printanzeigen-energiewende (accessed on 11 August 2021).
- 145. BMWi; BMU. Energy Concept for an Environmentally Sound, Reliable and Affordable Energy Supply; BMWi/BMU: Berlin, Germany, 2010.
- 146. Deutscher Bundestag. Gesetz zur Grundlegenden Reform Des Erneuerbare-Energien-Gesetzes und zur Änderung Weiterer der Energiewirtschaftsrechts; Deutscher Bundestag: Bonn, Germany, 2014; pp. 1066–1132.
- 147. Deutscher Bundestag. *Gesetz zur Neuregelung des Rechtsrahmens für die Frderung der Stromerzeugung aus Erneuerbaren Energien;* Deutscher Bundestag: Bonn, Germany, 2011; pp. 1–45.
- Purkus, A.; Gawel, E.; Deissenroth, M.; Nienhaus, K.; Wassermann, S. Market Integration of Renewable Energies through Direct Marketing—Lessons Learned from the German Market Premium Scheme. *Energy Sustain. Soc.* 2015, 5, 1–13. [CrossRef]
- 149. BMWi. Competition Instead of Fixed Funding Rates: Federal Network Agency Launches Auctions for Ground-Mounted PV Installations. Available online: https://www.bmwi.de/Redaktion/EN/Meldung/20150224-wettbewerb-statt-feste-foerdersaetzebundesnetzagentur-startet-photovoltaik-freiflaechenausschreibungen.html (accessed on 15 May 2021).
- 150. Deustcher Bundestag. Gesetz zur Förderung von Mieterstrom und zur Änderung Weiterer Vorschriften des Erneuerbare-Energien-Gesetzes; Deutscher Bundestag: Bonn, Germany, 2017; pp. 2532–2539.
- 151. BMWi. 2016 Revision Amending the Renewable-Energy-Sources-Act—Key Points; BMWi: Berlin/Bonn, Germany, 2015.
- 152. Haufe, M.C.; Ehrhart, K.M. Auctions for Renewable Energy Support—Suitability, Design, and First Lessons Learned. *Energy Policy* **2018**, 121, 217–224. [CrossRef]
- 153. Langer, K.; Decker, T.; Roosen, J.; Menrad, K. Factors Influencing Citizens' Acceptance and Non-Acceptance of Wind Energy in Germany. J. Clean. Prod. 2018, 175, 133–144. [CrossRef]
- 154. Lundberg, L. Auctions for All? Reviewing the German Wind Power Auctions in 2017. Energy Policy 2019, 128, 449–458. [CrossRef]
- 155. Bundesnetzagentur. Hintergrundpapier: Ergebnisse der Ersten Ausschreibungsrunde für Photovoltaik (PV)—Freiflächenanlagen Vom 15 April 2015; Deustcher Bundestag: Bonn, Germany, 2015.
- 156. Bundesnetzagentur. Hintergrundpapier: Ergebnisse der Zweiten Ausschreibungsrunde für Photovoltaik (PV)—Freiflächenanlagen Vom 1 August 2015; Deustcher Bundestag: Bonn, Germany, 2015.
- 157. Bundesnetzagentur. Hintergrundpapier: Vorläufi ge Ergebnisse der Dritten Ausschreibungsrunde für Photovoltaik (PV)—Freifl Ächenanlagen Vom 1. Dezember 2015; Deustcher Bundestag: Bonn, Germany, 2016.
- 158. Bundesnetzagentur. *Hintergrundpapier: Ergebnisse der Fünften Ausschreibungsrunde für Photovoltaik(PV)—Freiflächenanlagen Vom 1 August 2016;* Deustcher Bundestag: Bonn, Germany, 2016.
- 159. Bundesnetzagentur. Hintergrundpapier: Ergebnisse der Vierten Ausschreibungsrunde für Photovoltaik (PV)—Freiflächenanlagen Vom 1 April 2016; Deustcher Bundestag: Bonn, Germany, 2016.
- 160. Bundesnetzagentur. *Hintergrundpapier: Ergebnisse der Sechsten Ausschreibungsrunde für Photovoltaik(PV)*—*Freilächenanlagen Vom 1 Dezember 2016;* Deustcher Bundestag: Bonn, Germany, 2017.
- 161. Tews, K. The Crash of a Policy Pilot to Legally Define Community Energy. Evidence from the German Auction Scheme. *Sustainability* **2018**, *10*, 3397. [CrossRef]
- 162. Gsänger, S.; Karl, T. Community Wind Under the Auctions Model: A Critical Appraisal. In Accelerating the Transition to a 100% Renewable Energy Era; Uyar, T.S., Ed.; Springer International Publishing: Cham, Switzerland, 2020; pp. 233–257. ISBN 978-3-030-40738-4.
- 163. Amazo-Blanco, A.; Tiedemann, S.; Tay, S.; Bieri, M. Why There Is so Much Aggressive Bidding at Renewables Auctions—And What the Risks Are. Available online: https://energypost.eu/why-there-is-so-much-aggressive-bidding-at-renewables-auctionsand-what-the-risks-are/ (accessed on 15 May 2021).
- Bundesnetzagentur Beendete Ausschreibungen. Available online: https://www.bundesnetzagentur.de/DE/Sachgebiete/ElektrizitaetundGas/ Unternehmen\_Institutionen/Ausschreibungen/Wind\_Onshore/BeendeteAusschreibungen/BeendeteAusschreibungen\_node.html(accessed on 15 May 2021).
- 165. FA Wind Ausschreibungen für Windenergieanlagen. Available online: https://www.fachagentur-windenergie.de/themen/ ausschreibungen/ (accessed on 17 February 2022).
- Trutnevyte, E.; Strachan, N.; Dodds, P.E.; Pudjianto, D.; Strbac, G. Synergies and Trade-Offs between Governance and Costs in Electricity System Transition. *Energy Policy* 2015, 85, 170–181. [CrossRef]
- 167. Gawel, E.; Korte, K.; Tews, K. Distributional Challenges of Sustainability Policies—The Case of the German Energy Transition. *Sustainability* 2015, 7, 16599–16615. [CrossRef]
- 168. Busch, S.; Ortner, A. From National to Cross-Border Support of Renewable Electricity in the European Union. In *The European Dimension of Germany's Energy Transition: Opportunities and Conflicts*; Gawel, E., Strunz, S., Lehmann, P., Purkus, A., Eds.; Springer International Publishing: Cham, Switzerland, 2019; pp. 207–225. ISBN 978-3-030-03374-3.
- Winter, S.; Schlesewsky, L. The German Feed-in Tariff Revisited—An Empirical Investigation on Its Distributional Effects. *Energy Policy* 2019, 132, 344–356. [CrossRef]

- 170. Gawel, E.; Geißler, H.; Lehmann, P. Erschwinglichkeit Der Stromversorgung Und Förderung Erneuerbarer Energien—Eine Empirische Analyse Für Deutschland. In *Energie und Soziale Ungleichheit: Zur Gesellschaftlichen Dimension der Energiewende in Deutschland Und Europa*; Großmann, K., Schaffrin, A., Smigiel, C., Eds.; Springer Fachmedien Wiesbaden: Wiesbaden, Germany, 2017; pp. 319–346. ISBN 978-3-658-11723-8.
- 171. Frondel, M.; Sommer, S. Energiekostenbelastung Privater Haushalte—Das EEG Als Sozialpolitische Zeitbombe? In *List Forum für Wirtschafts-und Finanzpolitik*; 2014; Volume 40, pp. 382–402. [CrossRef]
- 172. Newell, P. Power Shift: The Global Political Economy of Energy Transitions; Cambridge University Press: Cambridge, UK, 2021; ISBN 9781108832854.
- 173. Finon, D.; Perez, Y. The Social Efficiency of Instruments of Promotion of Renewable Energies: A Transaction-Cost Perspective. *Ecol. Econ.* **2007**, *62*, 77–92. [CrossRef]
- 174. Finon, D. Pros and Cons of Alternative Policies Aimed at Promoting Renewables. In *An Efficient, Sustainable and Secure Supply of Energy for Europe Meeting the Challenge*; Riess, A., Kolev, A., Eds.; European Investment Bank (EIB): Luxembourg, 2007; pp. 110–133. ISBN 92-861-0596-8.
- 175. Tisdell, C.A. Competition, Diversity and Economic Performance: Processes, Complexities and Ecological Similarities; Edward Elgar Publishing: Cheltenham, UK, 2013; ISBN 978 1 84720 746 3.
- 176. Kalkbrenner, B.J.; Roosen, J. Citizens' Willingness to Participate in Local Renewable Energy Projects: The Role of Community and Trust in Germany. *Energy Res. Soc. Sci.* **2016**, *13*, 60–70. [CrossRef]
- 177. Pons-Seres de Brauwer, C.; Cohen, J.J. Analysing the Potential of Citizen-Financed Community Renewable Energy to Drive Europe's Low-Carbon Energy Transition. *Renew. Sustain. Energy Rev.* **2020**, *133*, 110300. [CrossRef]
- Warren, C.R.; McFadyen, M. Does Community Ownership Affect Public Attitudes to Wind Energy? A Case Study from South-West Scotland. Land Use Policy 2010, 27, 204–213. [CrossRef]